\begin{filecontents}{lutp95211.eps}
%!PS-Adobe-2.0
%%BoundingBox: 102 221 505 624
%%Title: hout.1 (Portrait A 4)
%%Pages: 1
%%Creator: HIGZ Version 1.21/10
%%CreationDate: 95/03/09   12.14
%%EndComments
/s {stroke} def /l {lineto} def /m {moveto} def /t {translate} def
/sw {stringwidth} def /r {rotate} def /rl {roll}  def /R {repeat} def
/d {rlineto} def /rm {rmoveto} def /gr {grestore} def /f {eofill} def
/c {setrgbcolor} def /lw {setlinewidth} def /sd {setdash} def
/cl {closepath} def /sf {scalefont setfont} def /black {0 setgray} def
/box {m dup 0 exch d exch 0 d 0 exch neg d cl} def
/C {initclip box clip newpath} def /NC {initclip} def
/bl {box s} def /bf {box f} def /Y { 0 exch d} def /X { 0 d} def 
/mp {newpath /y exch def /x exch def} def
/side {[w .77 mul w .23 mul] .385 w mul sd w 0 l currentpoint t -144 r} def
/mr {mp x y w2 0 360 arc} def /m24 {mr s} def /m20 {mr f} def
/mb {mp x y w2 add m w2 neg 0 d 0 w neg d w 0 d 0 w d cl} def
/mt {mp x y w2 add m w2 neg w neg d w 0 d cl} def
/m21 {mb f} def /m25 {mb s} def /m22 {mt f} def /m26 {mt s} def
/m23 {mp x y w2 sub m w2 w d w neg 0 d cl f} def
/m27 {mp x y w2 add m w3 neg w2 neg d w3 w2 neg d w3  w2 d cl s} def
/m28 {mp x w2 sub y w2 sub w3 add m w3 0 d 0 w3 neg d w3 0 d 0 w3 d w3 0 d
 0 w3 d w3 neg 0 d 0 w3 d w3 neg 0 d
 0 w3 neg d w3 neg 0 d cl s } def
/m29 {mp gsave x w2 sub y w2 add w3 sub m currentpoint t
 4 {side} repeat cl fill gr} def
/m30 {mp gsave x w2 sub y w2 add w3 sub m currentpoint t
 5 {side} repeat s gr} def
/m31 {mp x y w2 sub m 0 w d x w2 sub y m w 0 d
 x w2 sub y w2 add m w w neg d x w2 sub y w2
 sub m w w d s} def
/m2 {mp x y w2 sub m 0 w d x w2 sub y m w 0 d s} def
/m5 {mp x w2 sub y w2 sub m w w d x w2 sub y w2 add m w w neg d s} def
/DP {/PT exch def gsave 47.2 47.2 scale PT 1 eq { 16 16 1 [ 16 0 0 16 neg 0 16
] { < AA AA 55 55 AA AA 55 55 AA AA 55 55 AA AA 55 55 AA AA 55 55 AA AA 55 55
AA AA 55 55 AA AA 55 55 > } image } if PT 2 eq { 16 16 1 [ 16 0 0 16 neg 0 16
] { < BB BB EE EE BB BB EE EE BB BB EE EE BB BB EE EE BB BB EE EE BB BB EE EE
BB BB EE EE BB BB EE EE > } image } if PT 3 eq { 16 16 1 [ 16 0 0 16 neg 0 16
] { < FF FF BB BB FF FF EE EE FF FF BB BB FF FF EE EE FF FF BB BB FF FF EE EE
FF FF BB BB FF FF EE EE > } image } if PT 4 eq { 16 16 1 [ 16 0 0 16 neg 0 16
] { < DF DF BF BF 7F 7F FE FE FD FD FB FB F7 F7 EF EF DF DF BF BF 7F 7F FE FE
FD FD FB FB F7 F7 EF EF > } image } if PT 5 eq { 16 16 1 [ 16 0 0 16 neg 0 16
] { < 7F 7F BF B F DF DF EF EF F7 F7 FB FB FD FD FE FE 7F 7F BF BF DF DF EF
EF F7 F7 FB FB FD FD FE FE > } image } if PT 6 eq { 16 16 1 [ 16 0 0 16 neg 0
16 ] { < BB BB BB BB BB BB BB BB BB BB BB BB BB BB BB BB BB BB BB BB BB BB BB
BB BB BB BB BB BB BB BB BB > } image } if PT 7 eq { 16 16 1 [ 16 0 0 16 neg 0
16 ] { < FF FF FF FF FF FF 00 00 FF FF FF FF FF FF 00 00 FF FF FF FF FF FF 00
00 FF FF FF FF FF FF 00 00 > } image } if PT 8 eq { 16 16 1 [ 16 0 0 16 neg 0
16 ] { < EE EE 47 47 83 83 C5 C5 EE EE 5C 5C 38 38 74 74 EE EE 47 47 83 83 C5
C5 EE EE 5C 5C 38 38 74 74 > } image } if PT 9 eq { 16 16 1 [ 16 0 0 16 neg 0
16 ] { < EF EF EF EF D7 D7 38 38 FE FE FE FE 7D 7D 83 83 EF EF EF EF D7 D7 38
38 FE FE FE FE 7D 7D 83 83 > } image } if PT 10 eq { 16 16 1 [ 16 0 0 16 neg
0 16 ] { < EF EF EF EF EF EF 00 00 FE FE FE FE FE FE 00 00 EF EF EF EF EF EF
00 00 FE FE FE FE FE FE 00 00 > } image } if PT 11 eq { 16 16 1 [ 16 0 0 16
neg 0 16 ] { < F7 F7 B6 B6 D5 D5 E3 E3 D5 D5 B6 B6 F7 F7 FF FF 7F 7F 6B 6B 5D
5D 3E 3E 5D 5D 6B 6B 7F 7F FF FF > } image } if PT 12 eq { 16 16 1 [ 16 0 0
16 neg 0 16 ] { < E3 E3 DD DD BE BE BE BE BE BE DD DD E3 E3 FF FF 3E 3E DD DD
EB EB EB EB EB EB DD DD 3E 3E FF FF > } image } if PT 13 eq { 16 16 1 [ 16 0
0 16 neg 0 16 ] { < FE FE 7D 7D BB BB D7 D7 EF EF D7 D7 BB BB 7D 7D FE FE 7D
7D BB BB D7 D7 EF EF D7 D7 BB BB 7D 7D > } image } if PT 14 eq { 16 16 1 [ 16
0 0 16 neg 0 16 ] { < 00 00 EE EF EE EF EE EF 0E E0 EE EE EE EE EE EE 00 EE
FE EE FE EE FE EE 00 00 FE EF FE EF FE EF > } image } if PT 15 eq { 16 16 1 [
16 0 0 16 neg 0 16 ] { < DD DD AA AA DD DD FF FF 77 77 AA AA 77 77 FF FF DD
DD AA AA DD DD FF FF 77 77 AA AA 77 77 FF FF > } image } if PT 16 eq { 16 16
1 [ 16 0 0 16 neg 0 16 ] { < F1 F1 EE EE 1F 1F FF FF F1 F1 EE EE 1F 1F FF FF
F1 F1 EE EE 1F 1F FF FF F1 F1 EE EE 1F 1F FF FF > } image } if PT 17 eq { 16
16 1 [ 16 0 0 16 neg 0 16 ] { < EE EE DD DD BB BB FF FF EE EE DD DD BB BB FF
FF EE EE DD DD BB BB FF FF EE EE DD DD BB BB FF FF > } image } if PT 18 eq {
16 16 1 [ 16 0 0 16 neg 0 16 ] { < BB BB DD DD EE EE FF FF BB BB DD DD EE EE
FF FF BB BB DD DD EE EE FF FF BB BB DD DD EE EE FF FF > } image } if PT 19 eq
{ 16 16 1 [ 16 0 0 16 neg 0 16 ] { < 1F FC 67 F3 7B EF BD DE BD DE DE BD E6
B3 F8 0F E6 B3 DE BD BD DE BD DE 7B EF 67 F3 1F FC 7F FF > } image } if PT
20 eq { 16 16 1 [ 16 0 0 16 neg 0 16 ] { < DD DD EE EE EE EE EE EE DD DD BB
BB BB BB BB BB DD DD EE EE EE EE EE EE DD DD BB BB BB BB BB BB > } image }
if PT 21 eq { 16 16 1 [ 16 0 0 16 neg 0 16 ] { < 0E 0E EF EF EF EF EF EF E0
E0 FE FE FE FE FE FE 0E 0E EF EF EF EF EF EF E0 E0 FE FE FE FE FE FE > }
image } if PT 22 eq { 16 16 1 [ 16 0 0 16 neg 0 16 ] { < 70 70 F7 F7 F7 F7 F7
F7 07 07 7F 7F 7F 7F 7F 7F 70 70 F7 F7 F7 F7 F7 F7 07 07 7F 7F 7F 7F 7F 7F >
} image } if PT 23 eq { 16 16 1 [ 16 0 0 16 neg 0 16 ] { < AA AA 55 55 A9 A9
D1 D1 E1 E1 D1 D1 A9 A9 55 55 AA AA 55 55 A9 A9 D1 D1 E1 E1 D1 D1 A9 A9 55 55
> } image } if PT 24 eq { 16 16 1 [ 16 0 0 16 neg 0 16 ] { < FF FE FF FC EA
A8 D5 54 EA A8 D5 54 E8 28 D4 54 E8 E8 D4 D4 E8 EA 54 D5 A8 EA 54 D5 00 C0 00
80 > } image } if PT 25 eq { 16 16 1 [ 16 0 0 16 neg 0 16 ] { < FF FE FF FC
FF F8 FF F0 F0 00 F0 00 F0 20 F0 60 F0 E0 F1 E0 F3 E0 F0 00 E0 00 C0 00 80 00
00 00 > } image } if gr } def /FA { /PT exch def gsave clip 0 0 translate 1 1
54 { 1 sub 47.2 mul /Xcurr exch def 1 1 74 { 1 sub 47.2 mul /Ycurr exch def
gsave Xcurr Ycurr translate PT DP gr } for } for gr } def
/reencdict 24 dict def /ReEncode {reencdict begin /nco&na exch def
/nfnam exch def /basefontname exch def /basefontdict basefontname findfont def
/newfont basefontdict maxlength dict def basefontdict {exch dup /FID ne
{dup /Encoding eq {exch dup length array copy newfont 3 1 roll put} {exch
newfont 3 1 roll put} ifelse} {pop pop} ifelse } forall newfont
/FontName nfnam put nco&na aload pop nco&na length 2 idiv {newfont
/Encoding get 3 1 roll put} repeat nfnam newfont definefont pop end } def
/accvec [ 176 /agrave 181 /Agrave 190 /acircumflex 192 /Acircumflex
201 /adieresis 204 /Adieresis 209 /ccedilla 210 /Ccedilla 211 /eacute
212 /Eacute 213 /egrave 214 /Egrave 215 /ecircumflex 216 /Ecircumflex
217 /edieresis 218 /Edieresis 219 /icircumflex 220 /Icircumflex
221 /idieresis 222 /Idieresis 223 /ntilde 224 /Ntilde 226 /ocircumflex
228 /Ocircumflex 229 /odieresis 230 /Odieresis 231 /ucircumflex 236 /Ucircumflex
237 /udieresis 238 /Udieresis 239 /aring 242 /Aring 243 /ydieresis
244 /Ydieresis 246 /aacute 247 /Aacute 252 /ugrave 253 /Ugrave] def
/Times-Roman /Times-Roman accvec ReEncode
/Times-Italic /Times-Italic accvec ReEncode
/Times-Bold /Times-Bold accvec ReEncode
/Times-BoldItalic /Times-BoldItalic accvec ReEncode
/Helvetica /Helvetica accvec ReEncode
/Helvetica-Oblique /Helvetica-Oblique accvec ReEncode
/Helvetica-Bold /Helvetica-Bold accvec ReEncode
/Helvetica-BoldOblique /Helvetica-BoldOblique  accvec ReEncode
/Courier /Courier accvec ReEncode
/Courier-Oblique /Courier-Oblique accvec ReEncode
/Courier-Bold /Courier-Bold accvec ReEncode
/Courier-BoldOblique /Courier-BoldOblique accvec ReEncode
/oshow {gsave [] 0 sd true charpath stroke gr} def
/stwn { /fs exch def /fn exch def /text exch def fn findfont fs sf
 text sw pop xs add /xs exch def} def
/stwb { /fs exch def /fn exch def /nbas exch def /textf exch def
textf length /tlen exch def nbas tlen gt {/nbas tlen def} if
fn findfont fs sf textf dup length nbas sub nbas getinterval sw
pop neg xs add /xs exch def} def
/accspe [ 65 /plusminus 66 /bar 67 /existential 68 /universal
69 /exclam 70 /numbersign 71 /greater 72 /question 73 /integral
74 /colon 75 /semicolon 76 /less 77 /bracketleft 78 /bracketright
79 /greaterequal 80 /braceleft 81 /braceright 82 /radical
83 /spade 84 /heart 85 /diamond 86 /club 87 /lessequal
88 /multiply 89 /percent 90 /infinity 48 /circlemultiply 49 /circleplus
50 /emptyset 51 /lozenge 52 /bullet 53 /arrowright 54 /arrowup
55 /arrowleft 56 /arrowdown 57 /arrowboth 48 /degree 44 /comma 43 /plus
 45 /angle 42 /angleleft 47 /divide 61 /notequal 40 /equivalence 41 /second
 97 /approxequal 98 /congruent 99 /perpendicular 100 /partialdiff 101 /florin
 102 /intersection 103 /union 104 /propersuperset 105 /reflexsuperset
 106 /notsubset 107 /propersubset 108 /reflexsubset 109 /element 110 /notelement
 111 /gradient 112 /logicaland 113 /logicalor 114 /arrowdblboth
 115 /arrowdblleft 116 /arrowdblup 117 /arrowdblright 118 /arrowdbldown
 119 /ampersand 120 /omega1 121 /similar 122 /aleph ] def
/Symbol /Special accspe ReEncode
/Zone {/iy exch def /ix exch def  gsave ix 1 sub  2224 mul  1 iy sub  3144
 mul t} def
gsave 20 28 t .25 .25 scale  gsave
%%EndProlog
%%Page: number 1
 1 1 Zone
 gsave 0 0 t black [] 0 sd 1 lw 658 714 431 1663 bl 658 714 431 1663 C 431 1663
 m 658 X s NC 431 1663 m 714 Y s 465 1663 m -34 X s 448 1714 m -17 X s 448 1765
 m -17 X s 448 1816 m -17 X s 465 1867 m -34 X s 448 1918 m -17 X s 448 1969 m
 -17 X s 448 2020 m -17 X s 465 2071 m -34 X s 448 2122 m -17 X s 448 2173 m -17
 X s 448 2224 m -17 X s 465 2275 m -34 X s 465 2275 m -34 X s 448 2326 m -17 X s
 448 2377 m -17 X s 369 1679 m -5 -2 d -3 -4 d -1 -8 d -5 Y 1 -7 d 3 -5 d 5 -1 d
 3 X 5 1 d 3 5 d 1 7 d 5 Y -1 8 d -3 4 d -5 2 d -3 X cl s 324 1883 m -5 -2 d -3
 -4 d -1 -8 d -4 Y 1 -8 d 3 -4 d 5 -2 d 3 X 4 2 d 3 4 d 2 8 d 4 Y -2 8 d -3 4 d
 -4 2 d -3 X cl s 348 1854 m -2 -1 d 2 -2 d 1 2 d -1 1 d cl s 361 1875 m 2 Y 2 3
 d 1 1 d 3 2 d 7 X 3 -2 d 1 -1 d 2 -3 d -3 Y -2 -3 d -3 -5 d -15 -15 d 21 X s
 324 2087 m -5 -2 d -3 -4 d -1 -8 d -4 Y 1 -8 d 3 -4 d 5 -2 d 3 X 4 2 d 3 4 d 2
 8 d 4 Y -2 8 d -3 4 d -4 2 d -3 X cl s 348 2058 m -2 -1 d 2 -2 d 1 2 d -1 1 d
 cl s 375 2087 m -15 -21 d 23 X s 375 2087 m -32 Y s 324 2291 m -5 -1 d -3 -5 d
 -1 -8 d -4 Y 1 -8 d 3 -4 d 5 -2 d 3 X 4 2 d 3 4 d 2 8 d 4 Y -2 8 d -3 5 d -4 1
 d -3 X cl s 348 2262 m -2 -1 d 2 -2 d 1 2 d -1 1 d cl s 380 2287 m -2 3 d -4 1
 d -3 X -5 -1 d -3 -5 d -2 -8 d -7 Y 2 -6 d 3 -3 d 5 -2 d 1 X 5 2 d 3 3 d 1 4 d
 2 Y -1 4 d -3 3 d -5 2 d -1 X -5 -2 d -3 -3 d -2 -4 d s 431 1663 m 658 X s 431
 1697 m -34 Y s 497 1680 m -17 Y s 563 1680 m -17 Y s 628 1680 m -17 Y s 694
 1680 m -17 Y s 760 1697 m -34 Y s 826 1680 m -17 Y s 891 1680 m -17 Y s 957
 1680 m -17 Y s 1023 1680 m -17 Y s 1089 1697 m -34 Y s 429 1640 m -4 -1 d -3 -5
 d -2 -8 d -4 Y 2 -8 d 3 -4 d 4 -2 d 4 X 4 2 d 3 4 d 2 8 d 4 Y -2 8 d -3 5 d -4
 1 d -4 X cl s 736 1640 m -5 -1 d -3 -5 d -1 -8 d -4 Y 1 -8 d 3 -4 d 5 -2 d 3 X
 4 2 d 3 4 d 2 8 d 4 Y -2 8 d -3 5 d -4 1 d -3 X cl s 760 1611 m -2 -1 d 2 -2 d
 1 2 d -1 1 d cl s 790 1640 m -15 X -1 -14 d 1 2 d 5 1 d 4 X 5 -1 d 3 -3 d 1 -5
 d -3 Y -1 -4 d -3 -3 d -5 -2 d -4 X -5 2 d -1 1 d -2 3 d s 1083 1634 m 3 2 d 4
 4 d -32 Y s 658 714 431 1663 C 503 1866 m 34 Y s 503 1923 m 35 Y s /w 23
 def /w2 {w 2 div} def /w3 {w 3 div} def 503 1912 m20 608 1962 m 67 Y s 608 2052
 m 67 Y s 608 2040 m20 793 1906 m 52 Y s 793 1980 m 52 Y s 793 1969 m20 1701
 1702 261 721 C 635 2332 m20 689 2343 m -22 Y s 682 2343 m 15 X s 702 2343 m -22
 Y s 702 2343 m 10 X 3 -1 d 1 -1 d 1 -2 d -3 Y -1 -2 d -1 -2 d -3 -1 d -10 X s
 740 2338 m -1 2 d -2 2 d -2 1 d -5 X -2 -1 d -2 -2 d -1 -2 d -1 -3 d -6 Y 1 -3
 d 1 -2 d 2 -2 d 2 -1 d 5 X 2 1 d 2 2 d 1 2 d s 750 2322 m -1 -1 d -2 1 d 2 1 d
 1 -1 d -3 Y -1 -2 d -2 -1 d s 796 2343 m -2 -1 d -2 -2 d -1 -2 d -1 -3 d -6 Y 1
 -3 d 1 -2 d 2 -2 d 2 -1 d 4 X 3 1 d 2 2 d 1 2 d 1 3 d 6 Y -1 3 d -1 2 d -2 2 d
 -3 1 d -4 X cl s 799 2325 m 7 -7 d s 812 2346 m 1 Y 1 1 d 1 1 d 3 X 1 -1 d 1 -1
 d -1 Y -1 -1 d -1 -2 d -5 -5 d 7 X s 836 2334 m 20 X s 836 2327 m 20 X s 887
 2343 m -11 X -1 -9 d 1 1 d 3 1 d 4 X 3 -1 d 2 -3 d 1 -3 d -2 Y -1 -3 d -2 -2 d
 -3 -1 d -4 X -3 1 d -1 1 d -1 2 d s 898 2323 m -1 -1 d 1 -1 d 1 1 d -1 1 d cl s
 913 2343 m -3 -1 d -3 -3 d -1 -5 d -4 Y 1 -5 d 3 -3 d 3 -1 d 2 X 3 1 d 2 3 d 1
 5 d 4 Y -1 5 d -2 3 d -3 1 d -2 X cl s 942 2336 m -1 -4 d -2 -2 d -4 -1 d -1 X
 -3 1 d -2 2 d -1 4 d 1 Y 1 3 d 2 2 d 3 1 d 1 X 4 -1 d 2 -2 d 1 -4 d -6 Y -1 -5
 d -2 -3 d -4 -1 d -2 X -3 1 d -1 2 d s 976 2338 m -1 2 d -2 2 d -2 1 d -4 X -2
 -1 d -3 -2 d -1 -2 d -1 -3 d -6 Y 1 -3 d 1 -2 d 3 -2 d 2 -1 d 4 X 2 1 d 2 2 d 1
 2 d 3 Y s 971 2329 m 5 X s 983 2329 m 13 X 2 Y -1 3 d -1 1 d -2 1 d -4 X -2 -1
 d -2 -3 d -1 -3 d -2 Y 1 -3 d 2 -2 d 2 -1 d 4 X 2 1 d 2 2 d s 1000 2343 m 9 -22
 d s 1018 2343 m -9 -22 d s 1021 2346 m 2 Y 1 X 1 1 d 2 X 1 -1 d 1 X -3 Y -1 -2
 d -6 -5 d 8 X s 658 714 431 1663 C 459 1944 m 44 Y s 459 2011 m 44 Y s 459 2000
 m25 522 1916 m 31 Y s 522 1970 m 32 Y s 522 1959 m25 1701 1702 261 721 C 635
 2292 m25 689 2304 m -23 Y s 682 2304 m 15 X s 708 2304 m -3 -2 d -2 -2 d -1 -2
 d -1 -3 d -5 Y 1 -4 d 1 -2 d 2 -2 d 3 -1 d 4 X 2 1 d 2 2 d 1 2 d 1 4 d 5 Y -1 3
 d -1 2 d -2 2 d -2 2 d -4 X cl s 726 2304 m -23 Y s 726 2304 m 10 X 3 -2 d 1 -1
 d 1 -2 d -3 Y -1 -2 d -1 -1 d -3 -1 d -10 X s 754 2304 m -9 -23 d s 754 2304 m
 9 -23 d s 749 2288 m 10 X s 782 2304 m -15 -23 d s 767 2304 m 15 X s 767 2281 m
 15 X s 792 2282 m -1 -1 d -1 1 d 1 1 d 1 -1 d -2 Y -1 -2 d -1 -1 d s 817 2304 m
 -3 -2 d -2 -2 d -1 -2 d -1 -3 d -5 Y 1 -4 d 1 -2 d 2 -2 d 3 -1 d 4 X 2 1 d 2 2
 d 1 2 d 1 4 d 5 Y -1 3 d -1 2 d -2 2 d -2 2 d -4 X cl s 820 2285 m 6 -6 d s 833
 2307 m 1 Y 1 1 d 4 X 1 -1 d -3 Y -1 -2 d -6 -5 d 8 X s 857 2294 m 19 X s 857
 2287 m 19 X s 907 2304 m -10 X -2 -10 d 2 1 d 3 1 d 3 X 3 -1 d 2 -2 d 2 -3 d -3
 Y -2 -3 d -2 -2 d -3 -1 d -3 X -3 1 d -2 1 d -1 2 d s 918 2283 m -1 -1 d 1 -1 d
 1 1 d -1 1 d cl s 930 2299 m 2 1 d 3 4 d -23 Y s 955 2304 m -3 -2 d -3 -3 d -1
 -5 d -3 Y 1 -6 d 3 -3 d 3 -1 d 2 X 3 1 d 2 3 d 2 6 d 3 Y -2 5 d -2 3 d -3 2 d
 -2 X cl s 997 2298 m -1 2 d -2 2 d -2 2 d -5 X -2 -2 d -2 -2 d -1 -2 d -1 -3 d
 -5 Y 1 -4 d 1 -2 d 2 -2 d 2 -1 d 5 X 2 1 d 2 2 d 1 2 d 4 Y s 992 2290 m 5 X s
 1003 2290 m 13 X 2 Y -1 2 d -1 1 d -2 1 d -3 X -2 -1 d -2 -2 d -2 -3 d -3 Y 2
 -3 d 2 -2 d 2 -1 d 3 X 2 1 d 2 2 d s 1021 2304 m 8 -23 d s 1038 2304 m -9 -23 d
 s 1041 2307 m 1 1 d 1 Y 5 X -1 Y 1 -1 d -1 Y -1 -1 d -1 -2 d -5 -5 d 7 X s 658
 714 431 1663 C 466 1849 m 52 Y s 466 1924 m 53 Y s 466 1913 m23 512 1898 m 67 Y
 s 512 1987 m 68 Y s 512 1976 m23 593 1896 m 38 Y s 593 1957 m 37 Y s 593 1945
 m23 697 1935 m 52 Y s 697 2010 m 52 Y s 697 1999 m23 806 2025 m 75 Y s 806 2122
 m 75 Y s 806 2111 m23 921 1936 m 84 Y s 921 2043 m 83 Y s 921 2031 m23 1701
 1702 261 721 C 635 2253 m23 685 2264 m -23 Y s 685 2264 m 10 X 3 -1 d 1 -1 d 1
 -2 d -4 Y -1 -2 d -1 -1 d -3 -1 d -10 X s 708 2264 m -23 Y s 708 2241 m 12 X s
 726 2264 m -16 Y 1 -4 d 2 -2 d 3 -1 d 3 X 3 1 d 2 2 d 1 4 d 16 Y s 754 2264 m
 -23 Y s 746 2264 m 16 X s 772 2264 m -2 -1 d -2 -2 d -1 -3 d -1 -3 d -5 Y 1 -3
 d 1 -3 d 2 -2 d 2 -1 d 5 X 2 1 d 2 2 d 1 3 d 1 3 d 5 Y -1 3 d -1 3 d -2 2 d -2
 1 d -5 X cl s 793 2242 m -1 -1 d -1 1 d 1 1 d 1 -1 d -2 Y -1 -2 d -1 -1 d s 818
 2264 m -2 -1 d -3 -2 d -1 -3 d -1 -3 d -5 Y 1 -3 d 1 -3 d 3 -2 d 2 -1 d 4 X 2 1
 d 2 2 d 1 3 d 2 3 d 5 Y -2 3 d -1 3 d -2 2 d -2 1 d -4 X cl s 821 2245 m 6 -6 d
 s 834 2267 m 1 Y 1 1 d 1 1 d 2 X 1 -1 d 1 -1 d -3 Y -1 -1 d -6 -6 d 8 X s 858
 2254 m 19 X s 858 2248 m 19 X s 908 2264 m -10 X -1 -10 d 1 1 d 3 1 d 3 X 3 -1
 d 3 -2 d 1 -3 d -2 Y -1 -4 d -3 -2 d -3 -1 d -3 X -3 1 d -1 1 d -2 2 d s 919
 2243 m -1 -1 d 1 -1 d 1 1 d -1 1 d cl s 930 2264 m 12 X -7 -9 d 4 X 2 -1 d 1 -1
 d 1 -3 d -2 Y -1 -4 d -2 -2 d -3 -1 d -4 X -3 1 d -1 1 d -1 2 d s 956 2264 m -3
 -1 d -2 -3 d -2 -6 d -3 Y 2 -6 d 2 -3 d 3 -1 d 2 X 3 1 d 3 3 d 1 6 d 3 Y -1 6 d
 -3 3 d -3 1 d -2 X cl s 998 2258 m -1 3 d -2 2 d -2 1 d -5 X -2 -1 d -2 -2 d -1
 -3 d -1 -3 d -5 Y 1 -3 d 1 -3 d 2 -2 d 2 -1 d 5 X 2 1 d 2 2 d 1 3 d 3 Y s 993
 2250 m 5 X s 1005 2250 m 13 X 2 Y -2 2 d -1 1 d -2 1 d -3 X -2 -1 d -2 -2 d -1
 -3 d -2 Y 1 -4 d 2 -2 d 2 -1 d 3 X 2 1 d 3 2 d s 1022 2264 m 8 -23 d s 1039
 2264 m -9 -23 d s 1042 2267 m 1 1 d 1 Y 2 1 d 2 X 1 -1 d -1 Y 1 -1 d -1 Y -1 -1
 d -1 -1 d -5 -6 d 7 X s 658 714 431 1663 C 432 1936 m -27 Y 1 -7 d -14 Y 1 -6 d
 -6 Y 1 -4 d -4 Y 1 -1 d -3 Y 1 -2 d 1 -1 d -1 Y 1 -1 d 1 -1 d 2 X 2 1 d 1 X 2 2
 d 1 1 d 2 2 d 1 1 d 4 5 d 5 6 d 6 8 d 7 9 d 8 11 d 7 9 d 3 4 d 7 7 d 6 7 d 5 5
 d 7 8 d 7 6 d 6 5 d 3 3 d 7 5 d 6 5 d 7 4 d 4 3 d 7 3 d 6 3 d 7 3 d 4 2 d 6 2 d
 7 2 d 7 1 d 3 1 d 7 1 d 7 X 7 1 d 3 -1 d 8 X 6 -1 d 7 -2 d 3 X 7 -2 d 7 -2 d 6
 -3 d 3 -2 d 10 -4 d 5 -3 d 3 -2 d 4 -3 d 2 -2 d 2 -2 d 3 -3 d 3 -5 d 7 -10 d 3
 -3 d 2 -2 d 2 -2 d 1 -1 d 1 X 2 -1 d 2 -1 d 6 X 4 1 d 6 2 d 3 X 24 2 d 24 2 d
 23 3 d 19 2 d 5 1 d 18 3 d 6 1 d 17 3 d 7 2 d 17 4 d 6 1 d 20 5 d 4 1 d 24 7 d
 21 6 d 2 1 d 16 3 d 8 2 d 11 2 d 6 1 d 7 X 10 -1 d 3 X 4 -1 d 2 -1 d 2 -1 d 2
 -2 d 3 -2 d 4 -3 d 2 -4 d 3 -3 d 3 -4 d 3 -5 d 3 -5 d 2 -5 d 1 -2 d s
 [12 12] 0 sd 432 1915 m -26 Y 1 -8 d -14 Y 1 -7 d -6 Y 1 -6 d -5 Y 1 X -5 Y 1
 -4 d 1 -1 d 1 -3 d -1 Y 1 -3 d 1 -1 d 1 -2 d -1 Y 2 -2 d 1 -1 d 3 -2 d 3 -1 d 4
 -2 d 5 X 6 -1 d 7 1 d 8 X 10 1 d 18 2 d 23 2 d 48 X 23 -1 d 24 -3 d 24 -3 d 23
 -4 d 24 -5 d 24 -6 d 23 -6 d 24 -7 d 24 -7 d 23 -9 d 24 -8 d 24 -9 d 24 -10 d
 23 -9 d 24 -10 d 24 -10 d 23 -11 d 24 -9 d 24 -10 d 23 -9 d 18 -6 d 6 -2 d s
 432 1934 m -27 Y 1 -8 d -15 Y 1 -7 d -6 Y 1 -5 d -5 Y 1 -1 d -5 Y 1 -2 d 1 -1 d
 -3 Y 1 -1 d 1 -2 d 1 X 1 -2 d 2 -1 d 7 X 3 2 d 1 X 5 3 d 6 4 d 7 5 d 8 6 d 10 7
 d 10 7 d 8 5 d 9 5 d 10 6 d 4 2 d 10 5 d 11 5 d 3 1 d 11 4 d 13 5 d 11 4 d 12 3
 d 13 3 d 11 2 d 15 3 d 9 1 d 17 3 d 6 X 24 3 d 24 2 d 23 1 d 24 2 d 24 2 d 23 3
 d 20 2 d 4 1 d 18 3 d 6 1 d 18 3 d 6 2 d 18 4 d 5 1 d 24 6 d 24 7 d 22 6 d 1 1
 d 16 4 d 8 1 d 12 2 d 6 1 d 6 X 10 -1 d 4 X 3 -1 d 3 -1 d 2 -2 d 1 -1 d 3 -2 d
 4 -3 d 2 -4 d 3 -4 d 3 -4 d 3 -5 d 3 -5 d 3 -6 d s 432 1936 m -27 Y 1 -7 d -14
 Y 1 -6 d -6 Y 1 -4 d -4 Y 1 -1 d -3 Y 1 -2 d 1 -1 d -1 Y 1 -1 d 1 -1 d 2 X 2 1
 d 1 X 2 2 d 1 1 d 2 2 d 1 1 d 4 5 d 5 6 d 6 8 d 7 9 d 8 11 d 7 9 d 3 4 d 7 7 d
 6 7 d 5 5 d 7 8 d 7 6 d 6 5 d 3 3 d 7 5 d 6 5 d 7 4 d 4 3 d 7 3 d 6 3 d 7 3 d 4
 2 d 6 2 d 7 2 d 7 1 d 3 1 d 7 1 d 7 X 7 1 d 3 -1 d 8 X 6 -1 d 7 -2 d 3 X 7 -2 d
 7 -2 d 6 -3 d 3 -2 d 10 -4 d 5 -3 d 3 -2 d 4 -3 d 2 -2 d 2 -2 d 3 -3 d 3 -5 d 7
 -10 d 3 -3 d 2 -2 d 2 -2 d 1 -1 d 1 X 2 -1 d 2 -1 d 6 X 4 1 d 6 2 d 3 X 24 2 d
 24 2 d 23 3 d 19 2 d 5 1 d 18 3 d 6 1 d 17 3 d 7 2 d 17 4 d 6 1 d 20 5 d 4 1 d
 24 7 d 21 6 d 2 1 d 16 3 d 8 2 d 11 2 d 6 1 d 7 X 10 -1 d 3 X 4 -1 d 2 -1 d 2
 -1 d 2 -2 d 3 -2 d 4 -3 d 2 -4 d 3 -3 d 3 -4 d 3 -5 d 3 -5 d 2 -5 d 1 -2 d s
 1701 1702 261 721 C [] 0 sd 540 1739 m -2 2 d -3 1 d -5 X -3 -1 d -2 -2 d -2 Y
 1 -2 d 1 -2 d 2 -1 d 7 -2 d 2 -1 d 1 -1 d 1 -2 d -3 Y -2 -2 d -3 -2 d -5 X -3 2
 d -2 2 d s 560 1735 m -16 Y s 560 1731 m -3 2 d -2 2 d -3 X -2 -2 d -2 -2 d -1
 -3 d -2 Y 1 -3 d 2 -2 d 2 -2 d 3 X 2 2 d 3 2 d s 582 1739 m -2 2 d -3 1 d -5 X
 -3 -1 d -2 -2 d -2 Y 1 -2 d 1 -2 d 2 -1 d 7 -2 d 2 -1 d 1 -1 d 1 -2 d -3 Y -2
 -2 d -3 -2 d -5 X -3 2 d -2 2 d s 603 1738 m 2 1 d 3 3 d -23 Y s 622 1742 m -23
 Y s 622 1742 m 8 X 3 -1 d 2 -2 d 1 -2 d 1 -4 d -5 Y -1 -3 d -1 -2 d -2 -2 d -3
 -2 d -8 X s 662 1742 m -2 X -2 -1 d -1 -3 d -19 Y s 654 1735 m 7 X s 673 1735 m
 -2 -2 d -2 -2 d -1 -3 d -2 Y 1 -3 d 2 -2 d 2 -2 d 3 X 2 2 d 3 2 d 1 3 d 2 Y -1
 3 d -3 2 d -2 2 d -3 X cl s 689 1735 m -16 Y s 689 1728 m 1 3 d 2 2 d 3 2 d 3 X
 s 719 1742 m -2 -1 d -2 -2 d -1 -2 d -1 -4 d -5 Y 1 -3 d 1 -2 d 2 -2 d 2 -2 d 5
 X 2 2 d 2 2 d 1 2 d 1 3 d 5 Y -1 4 d -1 2 d -2 2 d -2 1 d -5 X cl s 723 1724 m
 6 -7 d s 736 1745 m 2 Y 1 X 1 1 d 2 X 1 -1 d 1 X -4 Y -2 -1 d -5 -6 d 8 X s 759
 1732 m 20 X s 759 1726 m 20 X s 810 1742 m -11 X -1 -10 d 1 1 d 4 2 d 3 X 3 -2
 d 2 -2 d 1 -3 d -2 Y -1 -3 d -2 -2 d -3 -2 d -3 X -4 2 d -1 1 d -1 2 d s 821
 1722 m -1 -1 d 1 -2 d 1 2 d -1 1 d cl s 831 1737 m 1 Y 1 2 d 1 1 d 2 1 d 4 X 2
 -1 d 2 -1 d 1 -2 d -2 Y -1 -3 d -3 -3 d -10 -11 d 15 X s 858 1742 m -4 -1 d -2
 -3 d -1 -6 d -3 Y 1 -5 d 2 -3 d 4 -2 d 2 X 3 2 d 2 3 d 1 5 d 3 Y -1 6 d -2 3 d
 -3 1 d -2 X cl s 900 1737 m -1 2 d -2 2 d -3 1 d -4 X -2 -1 d -2 -2 d -1 -2 d
 -1 -4 d -5 Y 1 -3 d 1 -2 d 2 -2 d 2 -2 d 4 X 3 2 d 2 2 d 1 2 d 3 Y s 894 1728 m
 6 X s 906 1728 m 13 X 2 Y -1 2 d -1 1 d -2 2 d -3 X -2 -2 d -3 -2 d -1 -3 d -2
 Y 1 -3 d 3 -2 d 2 -2 d 3 X 2 2 d 2 2 d s 924 1742 m 8 -23 d s 941 1742 m -9 -23
 d s 944 1745 m 1 Y 1 1 d 1 1 d 2 X 1 -1 d 1 X 1 -1 d -1 Y -1 -2 d -1 -1 d -5 -6
 d 7 X s NC 658 714 1259 1663 bl 658 714 1259 1663 C [12 12] 0 sd 1259 1663 m
 658 X s NC [] 0 sd 1259 1663 m 714 Y s 1293 1663 m -34 X s 1276 1714 m -17 X s
 1276 1765 m -17 X s 1276 1816 m -17 X s 1293 1867 m -34 X s 1276 1918 m -17 X s
 1276 1969 m -17 X s 1276 2020 m -17 X s 1293 2071 m -34 X s 1276 2122 m -17 X s
 1276 2173 m -17 X s 1276 2224 m -17 X s 1293 2275 m -34 X s 1293 2275 m -34 X s
 1276 2326 m -17 X s 1276 2377 m -17 X s 1197 1679 m -5 -2 d -3 -4 d -1 -8 d -5
 Y 1 -7 d 3 -5 d 5 -1 d 3 X 5 1 d 3 5 d 1 7 d 5 Y -1 8 d -3 4 d -5 2 d -3 X cl s
 1152 1883 m -5 -2 d -3 -4 d -1 -8 d -4 Y 1 -8 d 3 -4 d 5 -2 d 3 X 4 2 d 3 4 d 2
 8 d 4 Y -2 8 d -3 4 d -4 2 d -3 X cl s 1176 1854 m -2 -1 d 2 -2 d 1 2 d -1 1 d
 cl s 1189 1875 m 2 Y 2 3 d 1 1 d 3 2 d 6 X 4 -2 d 1 -1 d 2 -3 d -3 Y -2 -3 d -3
 -5 d -15 -15 d 21 X s 1152 2087 m -5 -2 d -3 -4 d -1 -8 d -4 Y 1 -8 d 3 -4 d 5
 -2 d 3 X 4 2 d 3 4 d 2 8 d 4 Y -2 8 d -3 4 d -4 2 d -3 X cl s 1176 2058 m -2 -1
 d 2 -2 d 1 2 d -1 1 d cl s 1203 2087 m -15 -21 d 23 X s 1203 2087 m -32 Y s
 1152 2291 m -5 -1 d -3 -5 d -1 -8 d -4 Y 1 -8 d 3 -4 d 5 -2 d 3 X 4 2 d 3 4 d 2
 8 d 4 Y -2 8 d -3 5 d -4 1 d -3 X cl s 1176 2262 m -2 -1 d 2 -2 d 1 2 d -1 1 d
 cl s 1208 2287 m -2 3 d -5 1 d -3 X -4 -1 d -3 -5 d -2 -8 d -7 Y 2 -6 d 3 -3 d
 4 -2 d 2 X 5 2 d 3 3 d 1 4 d 2 Y -1 4 d -3 3 d -5 2 d -2 X -4 -2 d -3 -3 d -2
 -4 d s 1259 1663 m 658 X s 1259 1697 m -34 Y s 1325 1680 m -17 Y s 1391 1680 m
 -17 Y s 1456 1680 m -17 Y s 1522 1680 m -17 Y s 1588 1697 m -34 Y s 1654 1680 m
 -17 Y s 1719 1680 m -17 Y s 1785 1680 m -17 Y s 1851 1680 m -17 Y s 1917 1697 m
 -34 Y s 1257 1640 m -4 -1 d -3 -5 d -2 -8 d -4 Y 2 -8 d 3 -4 d 4 -2 d 3 X 5 2 d
 3 4 d 2 8 d 4 Y -2 8 d -3 5 d -5 1 d -3 X cl s 1564 1640 m -5 -1 d -3 -5 d -1
 -8 d -4 Y 1 -8 d 3 -4 d 5 -2 d 3 X 4 2 d 3 4 d 2 8 d 4 Y -2 8 d -3 5 d -4 1 d
 -3 X cl s 1588 1611 m -2 -1 d 2 -2 d 1 2 d -1 1 d cl s 1618 1640 m -15 X -2 -14
 d 2 2 d 5 1 d 4 X 5 -1 d 3 -3 d 1 -5 d -3 Y -1 -4 d -3 -3 d -5 -2 d -4 X -5 2 d
 -2 1 d -1 3 d s 1911 1634 m 3 2 d 4 4 d -32 Y s 658 714 1259 1663 C 1331 1866 m
 34 Y s 1331 1923 m 35 Y s 1331 1912 m20 1436 1962 m 67 Y s 1436 2052 m 67 Y s
 1436 2040 m20 1621 1906 m 52 Y s 1621 1980 m 52 Y s 1621 1969 m20 1701 1702 261
 721 C 1463 2332 m20 1517 2343 m -22 Y s 1510 2343 m 15 X s 1530 2343 m -22 Y s
 1530 2343 m 10 X 3 -1 d 1 -1 d 1 -2 d -3 Y -1 -2 d -1 -2 d -3 -1 d -10 X s 1568
 2338 m -1 2 d -2 2 d -3 1 d -4 X -2 -1 d -2 -2 d -1 -2 d -1 -3 d -6 Y 1 -3 d 1
 -2 d 2 -2 d 2 -1 d 4 X 3 1 d 2 2 d 1 2 d s 1578 2322 m -1 -1 d -2 1 d 2 1 d 1
 -1 d -3 Y -1 -2 d -2 -1 d s 1624 2343 m -2 -1 d -2 -2 d -1 -2 d -1 -3 d -6 Y 1
 -3 d 1 -2 d 2 -2 d 2 -1 d 4 X 3 1 d 2 2 d 1 2 d 1 3 d 6 Y -1 3 d -1 2 d -2 2 d
 -3 1 d -4 X cl s 1627 2325 m 7 -7 d s 1640 2346 m 1 Y 1 1 d 1 1 d 3 X 1 -1 d 1
 -1 d -1 Y -1 -1 d -1 -2 d -5 -5 d 7 X s 1664 2334 m 19 X s 1664 2327 m 19 X s
 1715 2343 m -11 X -1 -9 d 1 1 d 3 1 d 3 X 4 -1 d 2 -3 d 1 -3 d -2 Y -1 -3 d -2
 -2 d -4 -1 d -3 X -3 1 d -1 1 d -1 2 d s 1726 2323 m -1 -1 d 1 -1 d 1 1 d -1 1
 d cl s 1741 2343 m -4 -1 d -2 -3 d -1 -5 d -4 Y 1 -5 d 2 -3 d 4 -1 d 2 X 3 1 d
 2 3 d 1 5 d 4 Y -1 5 d -2 3 d -3 1 d -2 X cl s 1770 2336 m -1 -4 d -2 -2 d -4
 -1 d -1 X -3 1 d -2 2 d -1 4 d 1 Y 1 3 d 2 2 d 3 1 d 1 X 4 -1 d 2 -2 d 1 -4 d
 -6 Y -1 -5 d -2 -3 d -4 -1 d -2 X -3 1 d -1 2 d s 1804 2338 m -1 2 d -2 2 d -2
 1 d -4 X -2 -1 d -3 -2 d -1 -2 d -1 -3 d -6 Y 1 -3 d 1 -2 d 3 -2 d 2 -1 d 4 X 2
 1 d 2 2 d 1 2 d 3 Y s 1799 2329 m 5 X s 1811 2329 m 13 X 2 Y -1 3 d -1 1 d -2 1
 d -4 X -2 -1 d -2 -3 d -1 -3 d -2 Y 1 -3 d 2 -2 d 2 -1 d 4 X 2 1 d 2 2 d s 1828
 2343 m 9 -22 d s 1845 2343 m -8 -22 d s 1849 2346 m 2 Y 1 X 1 1 d 2 X 1 -1 d 1
 X -3 Y -1 -2 d -6 -5 d 8 X s 658 714 1259 1663 C 1287 1944 m 44 Y s 1287 2011 m
 44 Y s 1287 2000 m25 1350 1916 m 31 Y s 1350 1970 m 32 Y s 1350 1959 m25 1701
 1702 261 721 C 1463 2292 m25 1517 2304 m -23 Y s 1510 2304 m 15 X s 1535 2304 m
 -2 -2 d -2 -2 d -1 -2 d -1 -3 d -5 Y 1 -4 d 1 -2 d 2 -2 d 2 -1 d 5 X 2 1 d 2 2
 d 1 2 d 1 4 d 5 Y -1 3 d -1 2 d -2 2 d -2 2 d -5 X cl s 1554 2304 m -23 Y s
 1554 2304 m 10 X 3 -2 d 1 -1 d 1 -2 d -3 Y -1 -2 d -1 -1 d -3 -1 d -10 X s 1582
 2304 m -9 -23 d s 1582 2304 m 9 -23 d s 1577 2288 m 10 X s 1610 2304 m -15 -23
 d s 1595 2304 m 15 X s 1595 2281 m 15 X s 1620 2282 m -1 -1 d -1 1 d 1 1 d 1 -1
 d -2 Y -1 -2 d -1 -1 d s 1645 2304 m -3 -2 d -2 -2 d -1 -2 d -1 -3 d -5 Y 1 -4
 d 1 -2 d 2 -2 d 3 -1 d 4 X 2 1 d 2 2 d 1 2 d 1 4 d 5 Y -1 3 d -1 2 d -2 2 d -2
 2 d -4 X cl s 1648 2285 m 6 -6 d s 1661 2307 m 1 Y 1 1 d 4 X 1 -1 d -3 Y -1 -2
 d -6 -5 d 8 X s 1685 2294 m 19 X s 1685 2287 m 19 X s 1735 2304 m -10 X -2 -10
 d 2 1 d 3 1 d 3 X 3 -1 d 2 -2 d 1 -3 d -3 Y -1 -3 d -2 -2 d -3 -1 d -3 X -3 1 d
 -2 1 d -1 2 d s 1746 2283 m -1 -1 d 1 -1 d 1 1 d -1 1 d cl s 1758 2299 m 2 1 d
 3 4 d -23 Y s 1783 2304 m -3 -2 d -3 -3 d -1 -5 d -3 Y 1 -6 d 3 -3 d 3 -1 d 2 X
 3 1 d 2 3 d 1 6 d 3 Y -1 5 d -2 3 d -3 2 d -2 X cl s 1825 2298 m -1 2 d -2 2 d
 -2 2 d -5 X -2 -2 d -2 -2 d -1 -2 d -1 -3 d -5 Y 1 -4 d 1 -2 d 2 -2 d 2 -1 d 5
 X 2 1 d 2 2 d 1 2 d 4 Y s 1820 2290 m 5 X s 1831 2290 m 13 X 2 Y -1 2 d -1 1 d
 -2 1 d -3 X -2 -1 d -2 -2 d -2 -3 d -3 Y 2 -3 d 2 -2 d 2 -1 d 3 X 2 1 d 2 2 d s
 1849 2304 m 8 -23 d s 1866 2304 m -9 -23 d s 1869 2307 m 1 1 d 1 Y 5 X -1 Y 1
 -1 d -1 Y -1 -1 d -1 -2 d -5 -5 d 7 X s 658 714 1259 1663 C 1294 1849 m 52 Y s
 1294 1924 m 53 Y s 1294 1913 m23 1340 1898 m 67 Y s 1340 1987 m 68 Y s 1340
 1976 m23 1421 1896 m 38 Y s 1421 1957 m 37 Y s 1421 1945 m23 1525 1935 m 52 Y s
 1525 2010 m 52 Y s 1525 1999 m23 1634 2025 m 75 Y s 1634 2122 m 75 Y s 1634
 2111 m23 1749 1936 m 84 Y s 1749 2043 m 83 Y s 1749 2031 m23 1701 1702 261 721
 C 1463 2253 m23 1513 2264 m -23 Y s 1513 2264 m 10 X 3 -1 d 1 -1 d 1 -2 d -4 Y
 -1 -2 d -1 -1 d -3 -1 d -10 X s 1535 2264 m -23 Y s 1535 2241 m 13 X s 1554
 2264 m -16 Y 1 -4 d 2 -2 d 3 -1 d 2 X 4 1 d 2 2 d 1 4 d 16 Y s 1582 2264 m -23
 Y s 1574 2264 m 15 X s 1600 2264 m -2 -1 d -2 -2 d -1 -3 d -1 -3 d -5 Y 1 -3 d
 1 -3 d 2 -2 d 2 -1 d 5 X 2 1 d 2 2 d 1 3 d 1 3 d 5 Y -1 3 d -1 3 d -2 2 d -2 1
 d -5 X cl s 1621 2242 m -1 -1 d -1 1 d 1 1 d 1 -1 d -2 Y -1 -2 d -1 -1 d s 1646
 2264 m -2 -1 d -3 -2 d -1 -3 d -1 -3 d -5 Y 1 -3 d 1 -3 d 3 -2 d 2 -1 d 4 X 2 1
 d 2 2 d 1 3 d 1 3 d 5 Y -1 3 d -1 3 d -2 2 d -2 1 d -4 X cl s 1649 2245 m 6 -6
 d s 1662 2267 m 1 Y 1 1 d 1 1 d 2 X 1 -1 d 1 -1 d -3 Y -1 -1 d -6 -6 d 8 X s
 1686 2254 m 19 X s 1686 2248 m 19 X s 1736 2264 m -10 X -1 -10 d 1 1 d 3 1 d 3
 X 3 -1 d 2 -2 d 2 -3 d -2 Y -2 -4 d -2 -2 d -3 -1 d -3 X -3 1 d -1 1 d -2 2 d s
 1747 2243 m -1 -1 d 1 -1 d 1 1 d -1 1 d cl s 1758 2264 m 12 X -7 -9 d 4 X 2 -1
 d 1 -1 d 1 -3 d -2 Y -1 -4 d -2 -2 d -4 -1 d -3 X -3 1 d -1 1 d -1 2 d s 1784
 2264 m -3 -1 d -2 -3 d -2 -6 d -3 Y 2 -6 d 2 -3 d 3 -1 d 2 X 3 1 d 2 3 d 2 6 d
 3 Y -2 6 d -2 3 d -3 1 d -2 X cl s 1826 2258 m -1 3 d -2 2 d -2 1 d -5 X -2 -1
 d -2 -2 d -1 -3 d -1 -3 d -5 Y 1 -3 d 1 -3 d 2 -2 d 2 -1 d 5 X 2 1 d 2 2 d 1 3
 d 3 Y s 1821 2250 m 5 X s 1833 2250 m 12 X 2 Y -1 2 d -1 1 d -2 1 d -3 X -2 -1
 d -2 -2 d -1 -3 d -2 Y 1 -4 d 2 -2 d 2 -1 d 3 X 2 1 d 2 2 d s 1850 2264 m 8 -23
 d s 1867 2264 m -9 -23 d s 1870 2267 m 1 1 d 1 Y 2 1 d 2 X 1 -1 d -1 Y 1 -1 d
 -1 Y -1 -1 d -1 -1 d -5 -6 d 7 X s 658 714 1259 1663 C 1260 2021 m -57 Y 1 -12
 d -10 Y 1 -10 d -9 Y 1 -6 d -7 Y 1 -2 d -6 Y 1 -3 d 1 -1 d -3 Y 1 -1 d 1 -2 d
 -1 Y 2 -1 d 3 X 2 1 d 4 2 d 4 4 d 5 5 d 5 6 d 7 7 d 9 9 d 10 10 d 10 10 d 8 6 d
 9 8 d 11 7 d 3 3 d 10 6 d 11 6 d 3 1 d 10 5 d 11 4 d 3 1 d 10 3 d 11 2 d 2 1 d
 11 1 d 10 1 d 13 X 10 -1 d 4 -1 d 11 -2 d 9 -2 d 3 -1 d 12 -5 d 6 -3 d 5 -3 d 1
 X 3 -3 d 4 -4 d 10 -13 d 3 -2 d 2 -2 d 2 -1 d 3 -1 d 3 X 4 1 d 8 3 d 4 2 d 1 X
 24 5 d 24 5 d 23 3 d 19 2 d 5 X 16 1 d 22 X 9 -1 d 13 -1 d 11 -2 d 13 -3 d 11
 -3 d 11 -4 d 10 -5 d 3 -1 d 10 -6 d 8 -5 d 5 -4 d 8 -6 d 7 -6 d 6 -7 d 3 -2 d 8
 -9 d 5 -7 d 5 -6 d 6 -9 d 8 -13 d 5 -8 d 3 -7 d 3 -7 d 3 -7 d 1 -3 d 4 -10 d 3
 -9 d 4 -12 d 3 -11 d 4 -13 d 3 -13 d 3 -13 d s [12 12] 0 sd 1260 1998 m -57 Y 1
 -12 d -12 Y 1 -11 d -10 Y 1 -8 d -9 Y 1 -1 d -8 Y 1 -6 d 1 -2 d 1 -6 d -1 Y 1
 -5 d -1 Y 2 -5 d -1 Y 2 -5 d 1 -1 d 2 -4 d -1 Y 4 -4 d 4 -5 d 5 -4 d 5 -4 d 7
 -3 d 9 -4 d 10 -4 d 18 -7 d 23 -8 d 24 -7 d 24 -8 d 23 -8 d 24 -7 d 24 -8 d 23
 -7 d 24 -8 d 24 -7 d 23 -7 d 24 -6 d 24 -7 d 23 -6 d 24 -6 d 24 -6 d 23 -5 d 24
 -4 d 24 -5 d 24 -3 d 23 -4 d 24 -3 d 24 -2 d 23 -1 d 24 -1 d s 1260 2018 m -59
 Y 1 -13 d -12 Y 1 -11 d -10 Y 1 -7 d -8 Y 1 -3 d -8 Y 1 -4 d 1 -3 d -3 Y 1 -3 d
 1 -3 d -2 Y 2 -3 d -2 Y 2 -2 d 1 -2 d 1 -1 d 1 -1 d 3 -2 d 1 -1 d 3 -1 d 4 X 2
 -1 d 4 1 d 1 X 7 1 d 9 1 d 10 2 d 18 4 d 15 3 d 8 1 d 15 2 d 9 1 d 17 1 d 7 X
 20 1 d 27 X 24 -1 d 71 X 18 1 d 5 X 17 2 d 7 1 d 15 1 d 9 2 d 14 2 d 9 2 d 14 3
 d 10 3 d 13 4 d 11 4 d 13 4 d 10 5 d 14 5 d 10 5 d 15 7 d 9 5 d 19 10 d 5 3 d
 23 13 d 18 10 d 6 3 d 12 6 d 7 3 d 5 1 d 10 3 d 4 1 d 6 X 2 -1 d 1 -1 d 3 -2 d
 4 -2 d 2 -2 d 3 -4 d 4 -3 d 3 -5 d 3 -4 d 2 -4 d s 1260 2019 m -57 Y 1 -13 d
 -11 Y 1 -10 d -10 Y 1 -6 d -7 Y 1 -3 d -6 Y 1 -4 d 1 -2 d -3 Y 1 -2 d 1 -2 d -2
 Y 1 -1 d 1 -1 d 2 -1 d 1 -1 d 6 X 3 1 d 1 1 d 4 2 d 1 1 d 5 4 d 7 5 d 9 6 d 10
 8 d 8 5 d 8 6 d 2 1 d 8 4 d 7 4 d 8 4 d 8 3 d 8 3 d 8 2 d 8 2 d 8 1 d 8 1 d 23
 X 9 -1 d 7 -1 d 8 -2 d 8 -2 d 8 -3 d 8 -3 d 8 -4 d 7 -3 d 7 -4 d 1 -1 d 11 -6 d
 4 -3 d 4 -3 d 4 -4 d 1 -1 d 24 -30 d 2 -1 d 2 X 3 -1 d 3 1 d 5 1 d 6 1 d 2 X 16
 2 d 8 1 d 14 1 d 10 2 d 13 2 d 10 2 d 13 3 d 11 3 d 12 4 d 12 4 d 12 4 d 11 5 d
 13 5 d 11 5 d 14 7 d 10 5 d 18 10 d 6 3 d 23 13 d 18 10 d 6 3 d 12 6 d 6 2 d 5
 2 d 1 X 10 3 d 4 1 d 5 X 3 -1 d 1 -1 d 3 -1 d 4 -3 d 2 -2 d 3 -3 d 3 -4 d 3 -4
 d 3 -4 d 3 -5 d s 1701 1702 261 721 C [] 0 sd 1368 1739 m -2 2 d -3 1 d -5 X -3
 -1 d -2 -2 d -2 Y 1 -2 d 1 -2 d 2 -1 d 7 -2 d 2 -1 d 1 -1 d 1 -2 d -3 Y -2 -2 d
 -3 -2 d -5 X -3 2 d -2 2 d s 1387 1735 m -16 Y s 1387 1731 m -2 2 d -2 2 d -3 X
 -2 -2 d -2 -2 d -1 -3 d -2 Y 1 -3 d 2 -2 d 2 -2 d 3 X 2 2 d 2 2 d s 1410 1739 m
 -2 2 d -3 1 d -5 X -3 -1 d -2 -2 d -2 Y 1 -2 d 1 -2 d 2 -1 d 7 -2 d 2 -1 d 1 -1
 d 1 -2 d -3 Y -2 -2 d -3 -2 d -5 X -3 2 d -2 2 d s 1431 1738 m 2 1 d 3 3 d -23
 Y s 1450 1742 m -23 Y s 1450 1742 m 9 -23 d s 1467 1742 m -8 -23 d s 1467 1742
 m -23 Y s 1493 1742 m -2 X -2 -1 d -1 -3 d -19 Y s 1485 1735 m 7 X s 1504 1735
 m -2 -2 d -2 -2 d -1 -3 d -2 Y 1 -3 d 2 -2 d 2 -2 d 3 X 3 2 d 2 2 d 1 3 d 2 Y
 -1 3 d -2 2 d -3 2 d -3 X cl s 1520 1735 m -16 Y s 1520 1728 m 1 3 d 3 2 d 2 2
 d 3 X s 1551 1742 m -3 -1 d -2 -2 d -1 -2 d -1 -4 d -5 Y 1 -3 d 1 -2 d 2 -2 d 3
 -2 d 4 X 2 2 d 2 2 d 1 2 d 1 3 d 5 Y -1 4 d -1 2 d -2 2 d -2 1 d -4 X cl s 1554
 1724 m 6 -7 d s 1567 1745 m 2 Y 1 X 1 1 d 2 X 1 -1 d 1 X -4 Y -1 -1 d -6 -6 d 8
 X s 1591 1732 m 19 X s 1591 1726 m 19 X s 1641 1742 m -10 X -2 -10 d 2 1 d 3 2
 d 3 X 3 -2 d 2 -2 d 2 -3 d -2 Y -2 -3 d -2 -2 d -3 -2 d -3 X -3 2 d -2 1 d -1 2
 d s 1652 1722 m -1 -1 d 1 -2 d 1 2 d -1 1 d cl s 1662 1737 m 1 Y 1 2 d 1 1 d 2
 1 d 5 X 2 -1 d 1 -1 d 1 -2 d -2 Y -1 -3 d -2 -3 d -11 -11 d 15 X s 1689 1742 m
 -3 -1 d -3 -3 d -1 -6 d -3 Y 1 -5 d 3 -3 d 3 -2 d 2 X 3 2 d 2 3 d 2 5 d 3 Y -2
 6 d -2 3 d -3 1 d -2 X cl s 1731 1737 m -1 2 d -2 2 d -2 1 d -5 X -2 -1 d -2 -2
 d -1 -2 d -1 -4 d -5 Y 1 -3 d 1 -2 d 2 -2 d 2 -2 d 5 X 2 2 d 2 2 d 1 2 d 3 Y s
 1726 1728 m 5 X s 1737 1728 m 13 X 2 Y -1 2 d -1 1 d -2 2 d -3 X -2 -2 d -2 -2
 d -2 -3 d -2 Y 2 -3 d 2 -2 d 2 -2 d 3 X 2 2 d 2 2 d s 1755 1742 m 8 -23 d s
 1772 1742 m -9 -23 d s 1775 1745 m 1 Y 1 1 d 1 1 d 3 X 1 -1 d 1 -1 d -1 Y -1 -2
 d -1 -1 d -5 -6 d 7 X s NC 658 714 431 835 bl 658 714 431 835 C [12 12] 0 sd
 431 835 m 658 X s NC [] 0 sd 431 835 m 714 Y s 465 835 m -34 X s 448 886 m -17
 X s 448 937 m -17 X s 448 988 m -17 X s 465 1039 m -34 X s 448 1090 m -17 X s
 448 1141 m -17 X s 448 1192 m -17 X s 465 1243 m -34 X s 448 1294 m -17 X s 448
 1345 m -17 X s 448 1396 m -17 X s 465 1447 m -34 X s 465 1447 m -34 X s 448
 1498 m -17 X s 369 851 m -5 -2 d -3 -4 d -1 -8 d -4 Y 1 -8 d 3 -5 d 5 -1 d 3 X
 5 1 d 3 5 d 1 8 d 4 Y -1 8 d -3 4 d -5 2 d -3 X cl s 324 1055 m -5 -2 d -3 -4 d
 -1 -8 d -4 Y 1 -8 d 3 -4 d 5 -2 d 3 X 4 2 d 3 4 d 2 8 d 4 Y -2 8 d -3 4 d -4 2
 d -3 X cl s 348 1026 m -2 -1 d 2 -2 d 1 2 d -1 1 d cl s 361 1047 m 2 Y 2 3 d 1
 1 d 3 2 d 7 X 3 -2 d 1 -1 d 2 -3 d -3 Y -2 -3 d -3 -5 d -15 -15 d 21 X s 324
 1259 m -5 -2 d -3 -4 d -1 -8 d -4 Y 1 -8 d 3 -4 d 5 -2 d 3 X 4 2 d 3 4 d 2 8 d
 4 Y -2 8 d -3 4 d -4 2 d -3 X cl s 348 1230 m -2 -1 d 2 -2 d 1 2 d -1 1 d cl s
 375 1259 m -15 -21 d 23 X s 375 1259 m -32 Y s 324 1463 m -5 -1 d -3 -5 d -1 -7
 d -5 Y 1 -8 d 3 -4 d 5 -2 d 3 X 4 2 d 3 4 d 2 8 d 5 Y -2 7 d -3 5 d -4 1 d -3 X
 cl s 348 1434 m -2 -1 d 2 -2 d 1 2 d -1 1 d cl s 380 1459 m -2 3 d -4 1 d -3 X
 -5 -1 d -3 -5 d -2 -7 d -8 Y 2 -6 d 3 -3 d 5 -2 d 1 X 5 2 d 3 3 d 1 4 d 2 Y -1
 4 d -3 4 d -5 1 d -1 X -5 -1 d -3 -4 d -2 -4 d s 431 835 m 658 X s 431 869 m
 -34 Y s 497 852 m -17 Y s 563 852 m -17 Y s 628 852 m -17 Y s 694 852 m -17 Y s
 760 869 m -34 Y s 826 852 m -17 Y s 891 852 m -17 Y s 957 852 m -17 Y s 1023
 852 m -17 Y s 1089 869 m -34 Y s 429 812 m -4 -1 d -3 -5 d -2 -8 d -4 Y 2 -8 d
 3 -4 d 4 -2 d 4 X 4 2 d 3 4 d 2 8 d 4 Y -2 8 d -3 5 d -4 1 d -4 X cl s 736 812
 m -5 -1 d -3 -5 d -1 -8 d -4 Y 1 -8 d 3 -4 d 5 -2 d 3 X 4 2 d 3 4 d 2 8 d 4 Y
 -2 8 d -3 5 d -4 1 d -3 X cl s 760 783 m -2 -1 d 2 -2 d 1 2 d -1 1 d cl s 790
 812 m -15 X -1 -14 d 1 2 d 5 2 d 4 X 5 -2 d 3 -3 d 1 -5 d -3 Y -1 -4 d -3 -3 d
 -5 -2 d -4 X -5 2 d -1 1 d -2 3 d s 1083 806 m 3 2 d 4 4 d -32 Y s 658 714 431
 835 C 503 1038 m 35 Y s 503 1095 m 35 Y s 503 1084 m20 608 1134 m 67 Y s 608
 1224 m 67 Y s 608 1212 m20 793 1078 m 52 Y s 793 1152 m 52 Y s 793 1141 m20
 1701 1702 261 721 C 635 1504 m20 689 1515 m -22 Y s 682 1515 m 15 X s 702 1515
 m -22 Y s 702 1515 m 10 X 3 -1 d 1 -1 d 1 -2 d -3 Y -1 -2 d -1 -2 d -3 -1 d -10
 X s 740 1510 m -1 2 d -2 2 d -2 1 d -5 X -2 -1 d -2 -2 d -1 -2 d -1 -3 d -6 Y 1
 -3 d 1 -2 d 2 -2 d 2 -1 d 5 X 2 1 d 2 2 d 1 2 d s 750 1494 m -1 -1 d -2 1 d 2 1
 d 1 -1 d -2 Y -1 -3 d -2 -1 d s 796 1515 m -2 -1 d -2 -2 d -1 -2 d -1 -3 d -6 Y
 1 -3 d 1 -2 d 2 -2 d 2 -1 d 4 X 3 1 d 2 2 d 1 2 d 1 3 d 6 Y -1 3 d -1 2 d -2 2
 d -3 1 d -4 X cl s 799 1497 m 7 -7 d s 812 1518 m 1 Y 1 1 d 1 1 d 3 X 1 -1 d 1
 -1 d -1 Y -1 -1 d -1 -2 d -5 -5 d 7 X s 836 1506 m 20 X s 836 1499 m 20 X s 887
 1515 m -11 X -1 -9 d 1 1 d 3 1 d 4 X 3 -1 d 2 -3 d 1 -3 d -2 Y -1 -3 d -2 -2 d
 -3 -1 d -4 X -3 1 d -1 1 d -1 2 d s 898 1495 m -1 -1 d 1 -1 d 1 1 d -1 1 d cl s
 913 1515 m -3 -1 d -3 -3 d -1 -5 d -4 Y 1 -5 d 3 -3 d 3 -1 d 2 X 3 1 d 2 3 d 1
 5 d 4 Y -1 5 d -2 3 d -3 1 d -2 X cl s 942 1508 m -1 -4 d -2 -2 d -4 -1 d -1 X
 -3 1 d -2 2 d -1 4 d 1 Y 1 3 d 2 2 d 3 1 d 1 X 4 -1 d 2 -2 d 1 -4 d -6 Y -1 -5
 d -2 -3 d -4 -1 d -2 X -3 1 d -1 2 d s 976 1510 m -1 2 d -2 2 d -2 1 d -4 X -2
 -1 d -3 -2 d -1 -2 d -1 -3 d -6 Y 1 -3 d 1 -2 d 3 -2 d 2 -1 d 4 X 2 1 d 2 2 d 1
 2 d 3 Y s 971 1501 m 5 X s 983 1501 m 13 X 2 Y -1 3 d -1 1 d -2 1 d -4 X -2 -1
 d -2 -3 d -1 -3 d -2 Y 1 -3 d 2 -2 d 2 -1 d 4 X 2 1 d 2 2 d s 1000 1515 m 9 -22
 d s 1018 1515 m -9 -22 d s 1021 1518 m 2 Y 1 X 1 1 d 2 X 1 -1 d 1 X -3 Y -1 -2
 d -6 -5 d 8 X s 658 714 431 835 C 459 1117 m 43 Y s 459 1183 m 44 Y s 459 1172
 m25 522 1088 m 31 Y s 522 1142 m 32 Y s 522 1131 m25 1701 1702 261 721 C 635
 1464 m25 689 1476 m -23 Y s 682 1476 m 15 X s 708 1476 m -3 -1 d -2 -3 d -1 -2
 d -1 -3 d -5 Y 1 -4 d 1 -2 d 2 -2 d 3 -1 d 4 X 2 1 d 2 2 d 1 2 d 1 4 d 5 Y -1 3
 d -1 2 d -2 3 d -2 1 d -4 X cl s 726 1476 m -23 Y s 726 1476 m 10 X 3 -1 d 1 -2
 d 1 -2 d -3 Y -1 -2 d -1 -1 d -3 -1 d -10 X s 754 1476 m -9 -23 d s 754 1476 m
 9 -23 d s 749 1460 m 10 X s 782 1476 m -15 -23 d s 767 1476 m 15 X s 767 1453 m
 15 X s 792 1454 m -1 -1 d -1 1 d 1 1 d 1 -1 d -2 Y -1 -2 d -1 -1 d s 817 1476 m
 -3 -1 d -2 -3 d -1 -2 d -1 -3 d -5 Y 1 -4 d 1 -2 d 2 -2 d 3 -1 d 4 X 2 1 d 2 2
 d 1 2 d 1 4 d 5 Y -1 3 d -1 2 d -2 3 d -2 1 d -4 X cl s 820 1457 m 6 -6 d s 833
 1479 m 1 Y 1 1 d 4 X 1 -1 d -3 Y -1 -2 d -6 -5 d 8 X s 857 1466 m 19 X s 857
 1459 m 19 X s 907 1476 m -10 X -2 -10 d 2 1 d 3 1 d 3 X 3 -1 d 2 -2 d 2 -3 d -3
 Y -2 -3 d -2 -2 d -3 -1 d -3 X -3 1 d -2 1 d -1 2 d s 918 1455 m -1 -1 d 1 -1 d
 1 1 d -1 1 d cl s 930 1471 m 2 1 d 3 4 d -23 Y s 955 1476 m -3 -1 d -3 -4 d -1
 -5 d -3 Y 1 -6 d 3 -3 d 3 -1 d 2 X 3 1 d 2 3 d 2 6 d 3 Y -2 5 d -2 4 d -3 1 d
 -2 X cl s 997 1470 m -1 2 d -2 3 d -2 1 d -5 X -2 -1 d -2 -3 d -1 -2 d -1 -3 d
 -5 Y 1 -4 d 1 -2 d 2 -2 d 2 -1 d 5 X 2 1 d 2 2 d 1 2 d 4 Y s 992 1462 m 5 X s
 1003 1462 m 13 X 2 Y -1 2 d -1 1 d -2 1 d -3 X -2 -1 d -2 -2 d -2 -3 d -3 Y 2
 -3 d 2 -2 d 2 -1 d 3 X 2 1 d 2 2 d s 1021 1476 m 8 -23 d s 1038 1476 m -9 -23 d
 s 1041 1479 m 1 1 d 1 Y 5 X -1 Y 1 -1 d -1 Y -1 -1 d -1 -2 d -5 -5 d 7 X s 658
 714 431 835 C 466 1021 m 53 Y s 466 1096 m 53 Y s 466 1085 m23 512 1070 m 67 Y
 s 512 1159 m 68 Y s 512 1148 m23 593 1069 m 37 Y s 593 1129 m 38 Y s 593 1118
 m23 697 1107 m 52 Y s 697 1182 m 52 Y s 697 1171 m23 806 1197 m 75 Y s 806 1294
 m 75 Y s 806 1283 m23 921 1108 m 84 Y s 921 1215 m 83 Y s 921 1203 m23 1701
 1702 261 721 C 635 1425 m23 685 1436 m -23 Y s 685 1436 m 10 X 3 -1 d 1 -1 d 1
 -2 d -4 Y -1 -2 d -1 -1 d -3 -1 d -10 X s 708 1436 m -23 Y s 708 1413 m 12 X s
 726 1436 m -16 Y 1 -4 d 2 -2 d 3 -1 d 3 X 3 1 d 2 2 d 1 4 d 16 Y s 754 1436 m
 -23 Y s 746 1436 m 16 X s 772 1436 m -2 -1 d -2 -2 d -1 -3 d -1 -3 d -5 Y 1 -3
 d 1 -3 d 2 -2 d 2 -1 d 5 X 2 1 d 2 2 d 1 3 d 1 3 d 5 Y -1 3 d -1 3 d -2 2 d -2
 1 d -5 X cl s 793 1414 m -1 -1 d -1 1 d 1 1 d 1 -1 d -2 Y -1 -2 d -1 -1 d s 818
 1436 m -2 -1 d -3 -2 d -1 -3 d -1 -3 d -5 Y 1 -3 d 1 -3 d 3 -2 d 2 -1 d 4 X 2 1
 d 2 2 d 1 3 d 2 3 d 5 Y -2 3 d -1 3 d -2 2 d -2 1 d -4 X cl s 821 1418 m 6 -7 d
 s 834 1439 m 1 Y 1 1 d 1 1 d 2 X 1 -1 d 1 -1 d -3 Y -1 -1 d -6 -6 d 8 X s 858
 1426 m 19 X s 858 1420 m 19 X s 908 1436 m -10 X -1 -10 d 1 1 d 3 1 d 3 X 3 -1
 d 3 -2 d 1 -3 d -2 Y -1 -4 d -3 -2 d -3 -1 d -3 X -3 1 d -1 1 d -2 3 d s 919
 1415 m -1 -1 d 1 -1 d 1 1 d -1 1 d cl s 930 1436 m 12 X -7 -9 d 4 X 2 -1 d 1 -1
 d 1 -3 d -2 Y -1 -4 d -2 -2 d -3 -1 d -4 X -3 1 d -1 1 d -1 3 d s 956 1436 m -3
 -1 d -2 -3 d -2 -6 d -3 Y 2 -5 d 2 -4 d 3 -1 d 2 X 3 1 d 3 4 d 1 5 d 3 Y -1 6 d
 -3 3 d -3 1 d -2 X cl s 998 1430 m -1 3 d -2 2 d -2 1 d -5 X -2 -1 d -2 -2 d -1
 -3 d -1 -3 d -5 Y 1 -3 d 1 -3 d 2 -2 d 2 -1 d 5 X 2 1 d 2 2 d 1 3 d 3 Y s 993
 1422 m 5 X s 1005 1422 m 13 X 2 Y -2 2 d -1 1 d -2 1 d -3 X -2 -1 d -2 -2 d -1
 -3 d -2 Y 1 -4 d 2 -2 d 2 -1 d 3 X 2 1 d 3 2 d s 1022 1436 m 8 -23 d s 1039
 1436 m -9 -23 d s 1042 1439 m 1 1 d 1 Y 2 1 d 2 X 1 -1 d -1 Y 1 -1 d -1 Y -1 -1
 d -1 -1 d -5 -6 d 7 X s 658 714 431 835 C 432 1045 m 5 Y 1 2 d 4 Y 1 3 d 3 Y 1
 3 d 1 4 d 4 Y 2 4 d 1 5 d 2 5 d 1 5 d 3 6 d 3 7 d 3 7 d 3 6 d 1 1 d 4 6 d 1 2 d
 4 7 d 2 2 d 5 7 d 2 2 d 6 7 d 2 3 d 6 6 d 4 4 d 7 7 d 8 7 d 3 2 d 8 6 d 9 6 d 6
 4 d 9 5 d 9 5 d 6 3 d 9 4 d 9 3 d 6 2 d 9 2 d 9 2 d 5 1 d 9 2 d 32 X 7 -1 d 8
 -2 d 7 -1 d 8 -3 d 11 -4 d 5 -2 d 4 -2 d 4 -3 d 3 -2 d 3 -4 d 11 -14 d 3 -2 d 2
 -1 d 2 -2 d 8 X 4 2 d 8 3 d 3 1 d 24 5 d 24 5 d 23 5 d 24 4 d 24 4 d 24 3 d 18
 2 d 5 X 17 1 d 7 1 d 24 X 14 -1 d 9 -1 d 14 -1 d 10 -2 d 11 -2 d 7 -2 d 6 -2 d
 10 -5 d 5 -2 d 4 -2 d 4 -3 d 5 -4 d 5 -5 d 5 -5 d 4 -6 d 5 -6 d s [12 12] 0 sd
 432 1043 m 4 Y 1 2 d 3 Y 1 2 d 3 Y 1 2 d 1 2 d 3 Y 2 3 d 1 3 d 2 3 d 1 4 d 3 3
 d 3 4 d 3 4 d 3 4 d 1 X 4 4 d 1 X 4 4 d 2 1 d 5 3 d 2 1 d 6 4 d 2 1 d 8 3 d 2 1
 d 8 3 d 10 4 d 10 3 d 13 4 d 20 5 d 4 1 d 24 5 d 23 6 d 24 5 d 24 5 d 23 6 d 24
 5 d 24 5 d 23 4 d 24 4 d 24 4 d 23 4 d 19 2 d 5 X 18 2 d 6 X 16 1 d 44 X 11 -1
 d 12 -1 d 12 -2 d 12 -2 d 10 -2 d 1 -1 d 12 -3 d 8 -2 d 4 -2 d 10 -4 d 7 -3 d 7
 -3 d 9 -6 d 5 -3 d 4 -3 d 4 -3 d 1 -1 d 5 -5 d 4 -5 d 5 -6 d 4 -5 d 5 -6 d 1 -3
 d s 432 1043 m 4 Y 1 2 d 3 Y 1 3 d 2 Y 1 2 d 1 3 d 3 Y 2 3 d 1 3 d 2 3 d 1 4 d
 3 4 d 3 4 d 3 4 d 4 4 d 5 5 d 5 4 d 1 1 d 5 4 d 2 X 7 5 d 1 X 8 5 d 2 1 d 10 4
 d 8 3 d 10 4 d 13 4 d 20 6 d 4 1 d 24 7 d 23 6 d 24 6 d 24 6 d 23 6 d 24 6 d 24
 5 d 23 6 d 24 5 d 24 5 d 23 5 d 24 4 d 24 4 d 20 3 d 4 X 18 2 d 5 X 17 1 d 7 1
 d 24 X 13 -1 d 10 -1 d 14 -1 d 10 -2 d 11 -2 d 7 -2 d 6 -2 d 10 -5 d 5 -2 d 4
 -2 d 4 -3 d 5 -4 d 5 -5 d 5 -5 d 4 -5 d 5 -7 d s 432 1045 m 5 Y 1 2 d 4 Y 1 3 d
 3 Y 1 3 d 1 4 d 4 Y 2 4 d 1 5 d 2 5 d 1 5 d 3 6 d 3 7 d 3 7 d 3 6 d 1 1 d 4 6 d
 1 2 d 4 7 d 2 2 d 5 7 d 2 2 d 6 7 d 2 3 d 6 6 d 4 4 d 7 7 d 8 7 d 3 2 d 8 6 d 9
 6 d 6 4 d 9 5 d 9 5 d 6 3 d 9 4 d 9 3 d 6 2 d 9 2 d 9 2 d 5 1 d 9 2 d 32 X 7 -1
 d 8 -2 d 7 -1 d 8 -3 d 11 -4 d 5 -2 d 4 -2 d 4 -3 d 3 -2 d 3 -4 d 11 -14 d 3 -2
 d 2 -1 d 2 -2 d 8 X 4 2 d 8 3 d 3 1 d 24 5 d 24 5 d 23 5 d 24 4 d 24 4 d 24 3 d
 18 2 d 5 X 17 1 d 7 1 d 24 X 14 -1 d 9 -1 d 14 -1 d 10 -2 d 11 -2 d 7 -2 d 6 -2
 d 10 -5 d 5 -2 d 4 -2 d 4 -3 d 5 -4 d 5 -5 d 5 -5 d 4 -6 d 5 -6 d s 1701 1702
 261 721 C [] 0 sd 540 911 m -2 2 d -3 1 d -5 X -3 -1 d -2 -2 d -2 Y 1 -2 d 1 -1
 d 2 -2 d 7 -2 d 2 -1 d 1 -1 d 1 -2 d -3 Y -2 -2 d -3 -2 d -5 X -3 2 d -2 2 d s
 560 907 m -16 Y s 560 903 m -3 3 d -2 1 d -3 X -2 -1 d -2 -3 d -1 -3 d -2 Y 1
 -3 d 2 -2 d 2 -2 d 3 X 2 2 d 3 2 d s 582 911 m -2 2 d -3 1 d -5 X -3 -1 d -2 -2
 d -2 Y 1 -2 d 1 -1 d 2 -2 d 7 -2 d 2 -1 d 1 -1 d 1 -2 d -3 Y -2 -2 d -3 -2 d -5
 X -3 2 d -2 2 d s 601 909 m 1 Y 1 2 d 1 1 d 2 1 d 4 X 2 -1 d 1 -1 d 2 -2 d -2 Y
 -2 -2 d -2 -4 d -10 -11 d 15 X s 622 914 m -23 Y s 622 914 m 8 X 3 -1 d 2 -2 d
 1 -2 d 1 -3 d -6 Y -1 -3 d -1 -2 d -2 -2 d -3 -2 d -8 X s 662 914 m -2 X -2 -1
 d -1 -3 d -19 Y s 654 907 m 7 X s 673 907 m -2 -1 d -2 -3 d -1 -3 d -2 Y 1 -3 d
 2 -2 d 2 -2 d 3 X 2 2 d 3 2 d 1 3 d 2 Y -1 3 d -3 3 d -2 1 d -3 X cl s 689 907
 m -16 Y s 689 900 m 1 3 d 2 3 d 3 1 d 3 X s 719 914 m -2 -1 d -2 -2 d -1 -2 d
 -1 -3 d -6 Y 1 -3 d 1 -2 d 2 -2 d 2 -2 d 5 X 2 2 d 2 2 d 1 2 d 1 3 d 6 Y -1 3 d
 -1 2 d -2 2 d -2 1 d -5 X cl s 723 896 m 6 -7 d s 736 917 m 2 Y 1 X 1 1 d 2 X 1
 -1 d 1 X -3 Y -2 -2 d -5 -6 d 8 X s 759 904 m 20 X s 759 898 m 20 X s 810 914 m
 -11 X -1 -10 d 1 2 d 4 1 d 3 X 3 -1 d 2 -3 d 1 -3 d -2 Y -1 -3 d -2 -2 d -3 -2
 d -3 X -4 2 d -1 1 d -1 2 d s 821 894 m -1 -1 d 1 -2 d 1 2 d -1 1 d cl s 831
 909 m 1 Y 1 2 d 1 1 d 2 1 d 4 X 2 -1 d 2 -1 d 1 -2 d -2 Y -1 -2 d -3 -4 d -10
 -11 d 15 X s 858 914 m -4 -1 d -2 -3 d -1 -6 d -3 Y 1 -5 d 2 -3 d 4 -2 d 2 X 3
 2 d 2 3 d 1 5 d 3 Y -1 6 d -2 3 d -3 1 d -2 X cl s 900 909 m -1 2 d -2 2 d -3 1
 d -4 X -2 -1 d -2 -2 d -1 -2 d -1 -3 d -6 Y 1 -3 d 1 -2 d 2 -2 d 2 -2 d 4 X 3 2
 d 2 2 d 1 2 d 3 Y s 894 900 m 6 X s 906 900 m 13 X 2 Y -1 2 d -1 2 d -2 1 d -3
 X -2 -1 d -3 -3 d -1 -3 d -2 Y 1 -3 d 3 -2 d 2 -2 d 3 X 2 2 d 2 2 d s 924 914 m
 8 -23 d s 941 914 m -9 -23 d s 944 917 m 1 Y 1 1 d 1 1 d 2 X 1 -1 d 1 X 1 -1 d
 -1 Y -1 -1 d -1 -2 d -5 -6 d 7 X s NC 658 714 1259 835 bl 658 714 1259 835 C
 [12 12] 0 sd 1259 835 m 658 X s NC [] 0 sd 1259 835 m 714 Y s 1293 835 m -34 X
 s 1276 886 m -17 X s 1276 937 m -17 X s 1276 988 m -17 X s 1293 1039 m -34 X s
 1276 1090 m -17 X s 1276 1141 m -17 X s 1276 1192 m -17 X s 1293 1243 m -34 X s
 1276 1294 m -17 X s 1276 1345 m -17 X s 1276 1396 m -17 X s 1293 1447 m -34 X s
 1293 1447 m -34 X s 1276 1498 m -17 X s 1197 851 m -5 -2 d -3 -4 d -1 -8 d -4 Y
 1 -8 d 3 -5 d 5 -1 d 3 X 5 1 d 3 5 d 1 8 d 4 Y -1 8 d -3 4 d -5 2 d -3 X cl s
 1152 1055 m -5 -2 d -3 -4 d -1 -8 d -4 Y 1 -8 d 3 -4 d 5 -2 d 3 X 4 2 d 3 4 d 2
 8 d 4 Y -2 8 d -3 4 d -4 2 d -3 X cl s 1176 1026 m -2 -1 d 2 -2 d 1 2 d -1 1 d
 cl s 1189 1047 m 2 Y 2 3 d 1 1 d 3 2 d 6 X 4 -2 d 1 -1 d 2 -3 d -3 Y -2 -3 d -3
 -5 d -15 -15 d 21 X s 1152 1259 m -5 -2 d -3 -4 d -1 -8 d -4 Y 1 -8 d 3 -4 d 5
 -2 d 3 X 4 2 d 3 4 d 2 8 d 4 Y -2 8 d -3 4 d -4 2 d -3 X cl s 1176 1230 m -2 -1
 d 2 -2 d 1 2 d -1 1 d cl s 1203 1259 m -15 -21 d 23 X s 1203 1259 m -32 Y s
 1152 1463 m -5 -1 d -3 -5 d -1 -7 d -5 Y 1 -8 d 3 -4 d 5 -2 d 3 X 4 2 d 3 4 d 2
 8 d 5 Y -2 7 d -3 5 d -4 1 d -3 X cl s 1176 1434 m -2 -1 d 2 -2 d 1 2 d -1 1 d
 cl s 1208 1459 m -2 3 d -5 1 d -3 X -4 -1 d -3 -5 d -2 -7 d -8 Y 2 -6 d 3 -3 d
 4 -2 d 2 X 5 2 d 3 3 d 1 4 d 2 Y -1 4 d -3 4 d -5 1 d -2 X -4 -1 d -3 -4 d -2
 -4 d s 1259 835 m 658 X s 1259 869 m -34 Y s 1325 852 m -17 Y s 1391 852 m -17
 Y s 1456 852 m -17 Y s 1522 852 m -17 Y s 1588 869 m -34 Y s 1654 852 m -17 Y s
 1719 852 m -17 Y s 1785 852 m -17 Y s 1851 852 m -17 Y s 1917 869 m -34 Y s
 1257 812 m -4 -1 d -3 -5 d -2 -8 d -4 Y 2 -8 d 3 -4 d 4 -2 d 3 X 5 2 d 3 4 d 2
 8 d 4 Y -2 8 d -3 5 d -5 1 d -3 X cl s 1564 812 m -5 -1 d -3 -5 d -1 -8 d -4 Y
 1 -8 d 3 -4 d 5 -2 d 3 X 4 2 d 3 4 d 2 8 d 4 Y -2 8 d -3 5 d -4 1 d -3 X cl s
 1588 783 m -2 -1 d 2 -2 d 1 2 d -1 1 d cl s 1618 812 m -15 X -2 -14 d 2 2 d 5 2
 d 4 X 5 -2 d 3 -3 d 1 -5 d -3 Y -1 -4 d -3 -3 d -5 -2 d -4 X -5 2 d -2 1 d -1 3
 d s 1911 806 m 3 2 d 4 4 d -32 Y s 658 714 1259 835 C 1331 1038 m 35 Y s 1331
 1095 m 35 Y s 1331 1084 m20 1436 1134 m 67 Y s 1436 1224 m 67 Y s 1436 1212 m20
 1621 1078 m 52 Y s 1621 1152 m 52 Y s 1621 1141 m20 1701 1702 261 721 C 1463
 1504 m20 1517 1515 m -22 Y s 1510 1515 m 15 X s 1530 1515 m -22 Y s 1530 1515 m
 10 X 3 -1 d 1 -1 d 1 -2 d -3 Y -1 -2 d -1 -2 d -3 -1 d -10 X s 1568 1510 m -1 2
 d -2 2 d -3 1 d -4 X -2 -1 d -2 -2 d -1 -2 d -1 -3 d -6 Y 1 -3 d 1 -2 d 2 -2 d
 2 -1 d 4 X 3 1 d 2 2 d 1 2 d s 1578 1494 m -1 -1 d -2 1 d 2 1 d 1 -1 d -2 Y -1
 -3 d -2 -1 d s 1624 1515 m -2 -1 d -2 -2 d -1 -2 d -1 -3 d -6 Y 1 -3 d 1 -2 d 2
 -2 d 2 -1 d 4 X 3 1 d 2 2 d 1 2 d 1 3 d 6 Y -1 3 d -1 2 d -2 2 d -3 1 d -4 X cl
 s 1627 1497 m 7 -7 d s 1640 1518 m 1 Y 1 1 d 1 1 d 3 X 1 -1 d 1 -1 d -1 Y -1 -1
 d -1 -2 d -5 -5 d 7 X s 1664 1506 m 19 X s 1664 1499 m 19 X s 1715 1515 m -11 X
 -1 -9 d 1 1 d 3 1 d 3 X 4 -1 d 2 -3 d 1 -3 d -2 Y -1 -3 d -2 -2 d -4 -1 d -3 X
 -3 1 d -1 1 d -1 2 d s 1726 1495 m -1 -1 d 1 -1 d 1 1 d -1 1 d cl s 1741 1515 m
 -4 -1 d -2 -3 d -1 -5 d -4 Y 1 -5 d 2 -3 d 4 -1 d 2 X 3 1 d 2 3 d 1 5 d 4 Y -1
 5 d -2 3 d -3 1 d -2 X cl s 1770 1508 m -1 -4 d -2 -2 d -4 -1 d -1 X -3 1 d -2
 2 d -1 4 d 1 Y 1 3 d 2 2 d 3 1 d 1 X 4 -1 d 2 -2 d 1 -4 d -6 Y -1 -5 d -2 -3 d
 -4 -1 d -2 X -3 1 d -1 2 d s 1804 1510 m -1 2 d -2 2 d -2 1 d -4 X -2 -1 d -3
 -2 d -1 -2 d -1 -3 d -6 Y 1 -3 d 1 -2 d 3 -2 d 2 -1 d 4 X 2 1 d 2 2 d 1 2 d 3 Y
 s 1799 1501 m 5 X s 1811 1501 m 13 X 2 Y -1 3 d -1 1 d -2 1 d -4 X -2 -1 d -2
 -3 d -1 -3 d -2 Y 1 -3 d 2 -2 d 2 -1 d 4 X 2 1 d 2 2 d s 1828 1515 m 9 -22 d s
 1845 1515 m -8 -22 d s 1849 1518 m 2 Y 1 X 1 1 d 2 X 1 -1 d 1 X -3 Y -1 -2 d -6
 -5 d 8 X s 658 714 1259 835 C 1287 1117 m 43 Y s 1287 1183 m 44 Y s 1287 1172
 m25 1350 1088 m 31 Y s 1350 1142 m 32 Y s 1350 1131 m25 1701 1702 261 721 C
 1463 1464 m25 1517 1476 m -23 Y s 1510 1476 m 15 X s 1535 1476 m -2 -1 d -2 -3
 d -1 -2 d -1 -3 d -5 Y 1 -4 d 1 -2 d 2 -2 d 2 -1 d 5 X 2 1 d 2 2 d 1 2 d 1 4 d
 5 Y -1 3 d -1 2 d -2 3 d -2 1 d -5 X cl s 1554 1476 m -23 Y s 1554 1476 m 10 X
 3 -1 d 1 -2 d 1 -2 d -3 Y -1 -2 d -1 -1 d -3 -1 d -10 X s 1582 1476 m -9 -23 d
 s 1582 1476 m 9 -23 d s 1577 1460 m 10 X s 1610 1476 m -15 -23 d s 1595 1476 m
 15 X s 1595 1453 m 15 X s 1620 1454 m -1 -1 d -1 1 d 1 1 d 1 -1 d -2 Y -1 -2 d
 -1 -1 d s 1645 1476 m -3 -1 d -2 -3 d -1 -2 d -1 -3 d -5 Y 1 -4 d 1 -2 d 2 -2 d
 3 -1 d 4 X 2 1 d 2 2 d 1 2 d 1 4 d 5 Y -1 3 d -1 2 d -2 3 d -2 1 d -4 X cl s
 1648 1457 m 6 -6 d s 1661 1479 m 1 Y 1 1 d 4 X 1 -1 d -3 Y -1 -2 d -6 -5 d 8 X
 s 1685 1466 m 19 X s 1685 1459 m 19 X s 1735 1476 m -10 X -2 -10 d 2 1 d 3 1 d
 3 X 3 -1 d 2 -2 d 1 -3 d -3 Y -1 -3 d -2 -2 d -3 -1 d -3 X -3 1 d -2 1 d -1 2 d
 s 1746 1455 m -1 -1 d 1 -1 d 1 1 d -1 1 d cl s 1758 1471 m 2 1 d 3 4 d -23 Y s
 1783 1476 m -3 -1 d -3 -4 d -1 -5 d -3 Y 1 -6 d 3 -3 d 3 -1 d 2 X 3 1 d 2 3 d 1
 6 d 3 Y -1 5 d -2 4 d -3 1 d -2 X cl s 1825 1470 m -1 2 d -2 3 d -2 1 d -5 X -2
 -1 d -2 -3 d -1 -2 d -1 -3 d -5 Y 1 -4 d 1 -2 d 2 -2 d 2 -1 d 5 X 2 1 d 2 2 d 1
 2 d 4 Y s 1820 1462 m 5 X s 1831 1462 m 13 X 2 Y -1 2 d -1 1 d -2 1 d -3 X -2
 -1 d -2 -2 d -2 -3 d -3 Y 2 -3 d 2 -2 d 2 -1 d 3 X 2 1 d 2 2 d s 1849 1476 m 8
 -23 d s 1866 1476 m -9 -23 d s 1869 1479 m 1 1 d 1 Y 5 X -1 Y 1 -1 d -1 Y -1 -1
 d -1 -2 d -5 -5 d 7 X s 658 714 1259 835 C 1294 1021 m 53 Y s 1294 1096 m 53 Y
 s 1294 1085 m23 1340 1070 m 67 Y s 1340 1159 m 68 Y s 1340 1148 m23 1421 1069 m
 37 Y s 1421 1129 m 38 Y s 1421 1118 m23 1525 1107 m 52 Y s 1525 1182 m 52 Y s
 1525 1171 m23 1634 1197 m 75 Y s 1634 1294 m 75 Y s 1634 1283 m23 1749 1108 m
 84 Y s 1749 1215 m 83 Y s 1749 1203 m23 1701 1702 261 721 C 1463 1425 m23 1513
 1436 m -23 Y s 1513 1436 m 10 X 3 -1 d 1 -1 d 1 -2 d -4 Y -1 -2 d -1 -1 d -3 -1
 d -10 X s 1535 1436 m -23 Y s 1535 1413 m 13 X s 1554 1436 m -16 Y 1 -4 d 2 -2
 d 3 -1 d 2 X 4 1 d 2 2 d 1 4 d 16 Y s 1582 1436 m -23 Y s 1574 1436 m 15 X s
 1600 1436 m -2 -1 d -2 -2 d -1 -3 d -1 -3 d -5 Y 1 -3 d 1 -3 d 2 -2 d 2 -1 d 5
 X 2 1 d 2 2 d 1 3 d 1 3 d 5 Y -1 3 d -1 3 d -2 2 d -2 1 d -5 X cl s 1621 1414 m
 -1 -1 d -1 1 d 1 1 d 1 -1 d -2 Y -1 -2 d -1 -1 d s 1646 1436 m -2 -1 d -3 -2 d
 -1 -3 d -1 -3 d -5 Y 1 -3 d 1 -3 d 3 -2 d 2 -1 d 4 X 2 1 d 2 2 d 1 3 d 1 3 d 5
 Y -1 3 d -1 3 d -2 2 d -2 1 d -4 X cl s 1649 1418 m 6 -7 d s 1662 1439 m 1 Y 1
 1 d 1 1 d 2 X 1 -1 d 1 -1 d -3 Y -1 -1 d -6 -6 d 8 X s 1686 1426 m 19 X s 1686
 1420 m 19 X s 1736 1436 m -10 X -1 -10 d 1 1 d 3 1 d 3 X 3 -1 d 2 -2 d 2 -3 d
 -2 Y -2 -4 d -2 -2 d -3 -1 d -3 X -3 1 d -1 1 d -2 3 d s 1747 1415 m -1 -1 d 1
 -1 d 1 1 d -1 1 d cl s 1758 1436 m 12 X -7 -9 d 4 X 2 -1 d 1 -1 d 1 -3 d -2 Y
 -1 -4 d -2 -2 d -4 -1 d -3 X -3 1 d -1 1 d -1 3 d s 1784 1436 m -3 -1 d -2 -3 d
 -2 -6 d -3 Y 2 -5 d 2 -4 d 3 -1 d 2 X 3 1 d 2 4 d 2 5 d 3 Y -2 6 d -2 3 d -3 1
 d -2 X cl s 1826 1430 m -1 3 d -2 2 d -2 1 d -5 X -2 -1 d -2 -2 d -1 -3 d -1 -3
 d -5 Y 1 -3 d 1 -3 d 2 -2 d 2 -1 d 5 X 2 1 d 2 2 d 1 3 d 3 Y s 1821 1422 m 5 X
 s 1833 1422 m 12 X 2 Y -1 2 d -1 1 d -2 1 d -3 X -2 -1 d -2 -2 d -1 -3 d -2 Y 1
 -4 d 2 -2 d 2 -1 d 3 X 2 1 d 2 2 d s 1850 1436 m 8 -23 d s 1867 1436 m -9 -23 d
 s 1870 1439 m 1 1 d 1 Y 2 1 d 2 X 1 -1 d -1 Y 1 -1 d -1 Y -1 -1 d -1 -1 d -5 -6
 d 7 X s 658 714 1259 835 C 1260 1022 m 8 Y 1 3 d 2 Y 1 4 d 3 Y 1 4 d 1 5 d 4 Y
 2 6 d 1 6 d 1 6 d 2 7 d 3 8 d 2 7 d 1 Y 3 7 d 1 2 d 3 7 d 1 3 d 3 7 d 2 3 d 3 7
 d 2 3 d 5 7 d 2 4 d 6 7 d 3 4 d 5 7 d 5 5 d 7 7 d 7 6 d 4 3 d 6 6 d 9 5 d 8 6 d
 9 5 d 10 4 d 5 2 d 10 4 d 10 3 d 4 1 d 10 2 d 10 2 d 3 1 d 10 1 d 24 X 10 -1 d
 4 X 11 -2 d 8 -2 d 4 -2 d 12 -3 d 5 -3 d 5 -2 d 2 -1 d 3 -3 d 3 -3 d 11 -13 d 2
 -2 d 3 -2 d 2 -1 d 5 X 3 1 d 4 1 d 9 4 d 2 1 d 24 7 d 20 5 d 4 1 d 16 3 d 7 2 d
 14 2 d 10 1 d 12 1 d 35 X 10 -2 d 10 -1 d 4 -1 d 9 -3 d 10 -2 d 5 -2 d 9 -4 d 8
 -4 d 7 -4 d 8 -5 d 8 -5 d 7 -5 d 8 -7 d 7 -6 d 7 -7 d 2 -2 d 8 -8 d 6 -8 d 6 -7
 d 4 -5 d 7 -11 d 6 -8 d 4 -7 d 4 -7 d 2 -4 d 5 -9 d 5 -10 d 5 -11 d 5 -11 d 4
 -10 d s [12 12] 0 sd 1260 1020 m 5 Y 1 1 d 2 Y 1 2 d 2 Y 1 3 d 1 2 d 3 Y 2 3 d
 1 3 d 1 4 d 2 3 d 3 4 d 2 4 d 3 4 d 1 X 3 4 d 1 1 d 4 3 d 1 1 d 4 3 d 1 1 d 5 3
 d 2 1 d 6 2 d 3 2 d 6 2 d 4 1 d 8 1 d 10 2 d 8 1 d 39 X 24 -1 d 23 -2 d 24 -1 d
 24 -1 d 65 X 6 1 d 17 1 d 6 1 d 18 2 d 6 X 17 3 d 7 1 d 17 3 d 6 2 d 18 3 d 6 2
 d 18 5 d 6 1 d 19 6 d 4 1 d 21 7 d 3 1 d 24 8 d 24 9 d 23 9 d 24 8 d 19 7 d 5 2
 d 13 4 d 6 1 d 4 1 d 6 1 d 7 X 5 -1 d 6 -1 d s 1260 1020 m 5 Y 1 1 d 2 Y 1 2 d
 2 Y 1 3 d 1 3 d 2 Y 2 4 d 1 3 d 1 3 d 2 4 d 3 4 d 2 5 d 4 4 d 4 5 d 5 4 d 5 4 d
 1 Y 6 3 d 1 1 d 7 3 d 2 1 d 7 3 d 3 1 d 9 2 d 9 2 d 9 2 d 14 1 d 12 X 12 1 d 24
 -1 d 23 X 24 -1 d 47 X 24 1 d 20 1 d 4 1 d 19 2 d 4 X 19 3 d 5 1 d 18 4 d 6 1 d
 18 4 d 5 2 d 19 5 d 5 2 d 18 6 d 6 2 d 18 7 d 5 2 d 20 8 d 4 2 d 21 9 d 3 2 d
 24 12 d 23 12 d 24 12 d 21 11 d 3 2 d 14 7 d 5 2 d 4 1 d 7 2 d 7 1 d 10 X s
 1260 1021 m 7 Y 1 2 d 2 Y 1 3 d 3 Y 1 3 d 1 4 d 4 Y 2 4 d 1 5 d 1 5 d 2 6 d 3 6
 d 2 7 d 4 7 d 3 6 d 1 2 d 4 6 d 1 2 d 4 6 d 1 2 d 5 6 d 2 3 d 6 6 d 3 3 d 6 5 d
 4 4 d 7 5 d 7 5 d 4 2 d 7 4 d 8 3 d 8 4 d 9 3 d 9 2 d 6 1 d 9 1 d 10 1 d 14 X
 10 -1 d 4 X 10 -2 d 10 -2 d 4 -1 d 10 -3 d 10 -3 d 4 -2 d 10 -4 d 9 -5 d 4 -2 d
 12 -7 d 6 -4 d 4 -3 d 2 -2 d 24 -28 d 2 -1 d 3 -1 d 3 X 5 1 d 8 3 d 2 X 19 3 d
 5 1 d 18 4 d 6 1 d 18 4 d 5 2 d 18 5 d 6 2 d 18 6 d 6 2 d 18 7 d 5 2 d 20 8 d 4
 2 d 20 9 d 4 2 d 24 12 d 23 12 d 24 12 d 21 11 d 3 2 d 13 7 d 6 2 d 4 1 d 7 2 d
 7 1 d 10 X s 1701 1702 261 721 C [] 0 sd 1368 911 m -2 2 d -3 1 d -5 X -3 -1 d
 -2 -2 d -2 Y 1 -2 d 1 -1 d 2 -2 d 7 -2 d 2 -1 d 1 -1 d 1 -2 d -3 Y -2 -2 d -3
 -2 d -5 X -3 2 d -2 2 d s 1387 907 m -16 Y s 1387 903 m -2 3 d -2 1 d -3 X -2
 -1 d -2 -3 d -1 -3 d -2 Y 1 -3 d 2 -2 d 2 -2 d 3 X 2 2 d 2 2 d s 1410 911 m -2
 2 d -3 1 d -5 X -3 -1 d -2 -2 d -2 Y 1 -2 d 1 -1 d 2 -2 d 7 -2 d 2 -1 d 1 -1 d
 1 -2 d -3 Y -2 -2 d -3 -2 d -5 X -3 2 d -2 2 d s 1429 909 m 1 Y 1 2 d 1 1 d 2 1
 d 4 X 2 -1 d 1 -1 d 2 -2 d -2 Y -2 -2 d -2 -4 d -11 -11 d 16 X s 1450 914 m -23
 Y s 1450 914 m 9 -23 d s 1467 914 m -8 -23 d s 1467 914 m -23 Y s 1493 914 m -2
 X -2 -1 d -1 -3 d -19 Y s 1485 907 m 7 X s 1504 907 m -2 -1 d -2 -3 d -1 -3 d
 -2 Y 1 -3 d 2 -2 d 2 -2 d 3 X 3 2 d 2 2 d 1 3 d 2 Y -1 3 d -2 3 d -3 1 d -3 X
 cl s 1520 907 m -16 Y s 1520 900 m 1 3 d 3 3 d 2 1 d 3 X s 1551 914 m -3 -1 d
 -2 -2 d -1 -2 d -1 -3 d -6 Y 1 -3 d 1 -2 d 2 -2 d 3 -2 d 4 X 2 2 d 2 2 d 1 2 d
 1 3 d 6 Y -1 3 d -1 2 d -2 2 d -2 1 d -4 X cl s 1554 896 m 6 -7 d s 1567 917 m
 2 Y 1 X 1 1 d 2 X 1 -1 d 1 X -3 Y -1 -2 d -6 -6 d 8 X s 1591 904 m 19 X s 1591
 898 m 19 X s 1641 914 m -10 X -2 -10 d 2 2 d 3 1 d 3 X 3 -1 d 2 -3 d 2 -3 d -2
 Y -2 -3 d -2 -2 d -3 -2 d -3 X -3 2 d -2 1 d -1 2 d s 1652 894 m -1 -1 d 1 -2 d
 1 2 d -1 1 d cl s 1662 909 m 1 Y 1 2 d 1 1 d 2 1 d 5 X 2 -1 d 1 -1 d 1 -2 d -2
 Y -1 -2 d -2 -4 d -11 -11 d 15 X s 1689 914 m -3 -1 d -3 -3 d -1 -6 d -3 Y 1 -5
 d 3 -3 d 3 -2 d 2 X 3 2 d 2 3 d 2 5 d 3 Y -2 6 d -2 3 d -3 1 d -2 X cl s 1731
 909 m -1 2 d -2 2 d -2 1 d -5 X -2 -1 d -2 -2 d -1 -2 d -1 -3 d -6 Y 1 -3 d 1
 -2 d 2 -2 d 2 -2 d 5 X 2 2 d 2 2 d 1 2 d 3 Y s 1726 900 m 5 X s 1737 900 m 13 X
 2 Y -1 2 d -1 2 d -2 1 d -3 X -2 -1 d -2 -3 d -2 -3 d -2 Y 2 -3 d 2 -2 d 2 -2 d
 3 X 2 2 d 2 2 d s 1755 914 m 8 -23 d s 1772 914 m -9 -23 d s 1775 917 m 1 Y 1 1
 d 1 1 d 3 X 1 -1 d 1 -1 d -1 Y -1 -1 d -1 -2 d -5 -6 d 7 X s gr 1701 2552 261
 296 C NC
showpage gr
%%Trailer
gr gr
%%EOF
\end{filecontents}

\documentclass[12pt]{article}
\usepackage{epsfig}

%define page size
\setlength{\textheight}{245mm}
\setlength{\topmargin}{-5mm}
\setlength{\headheight}{0mm}
\setlength{\headsep}{0mm}
\setlength{\footskip}{10mm}
\setlength{\textwidth}{160mm}
\setlength{\oddsidemargin}{0mm}
\setlength{\evensidemargin}{0mm}
 
%line spacing and eqnarray column spacing
\setlength{\arraycolsep}{0.5mm}

%shorthand for mathroman
\newcommand{\mrm}[1]{\mathrm{#1}}
 
%some frequent math symbols
\newcommand{\gtrsim}{\raisebox{-0.8mm}%
{\hspace{1mm}$\stackrel{>}{\sim}$\hspace{1mm}}}
\newcommand{\lessim}{\raisebox{-0.8mm}%
{\hspace{1mm}$\stackrel{<}{\sim}$\hspace{1mm}}}
 
%some frequent physics symbols
\newcommand{\alphas}{\alpha_{\mrm{s}}}
\newcommand{\alphaem}{\alpha_{\mrm{em}}}
\newcommand{\ssintw}{\sin^2 \! \theta_W}
\newcommand{\scostw}{\cos^2 \! \theta_W}
\newcommand{\pT}{p_{\perp}}
\newcommand{\kT}{k_{\perp}}
\newcommand{\pTmin}{p_{\perp\mrm{min}}}
\newcommand{\br}[1]{\overline{#1}}
\newcommand{\pom}{\mrm{I}\!\mrm{P}}
\newcommand{\reg}{\mrm{I}\!\mrm{R}}
 
%roman names for particles in math mode
\renewcommand{\a}{\mrm{a}}
\renewcommand{\b}{\mrm{b}}
\renewcommand{\c}{\mrm{c}}
\renewcommand{\d}{\mrm{d}}
\newcommand{\e}{\mrm{e}}
\newcommand{\f}{\mrm{f}}
\newcommand{\g}{\mrm{g}}
\newcommand{\hrm}{\mrm{h}}
\newcommand{\lrm}{\mrm{l}}
\newcommand{\n}{\mrm{n}}
\newcommand{\p}{\mrm{p}}
\newcommand{\q}{\mrm{q}}
\newcommand{\s}{\mrm{s}}
\renewcommand{\t}{\mrm{t}}
\renewcommand{\u}{\mrm{u}}
\newcommand{\A}{\mrm{A}}
\newcommand{\B}{\mrm{B}}
\newcommand{\C}{\mrm{C}}
\newcommand{\D}{\mrm{D}}
\newcommand{\F}{\mrm{F}}
\renewcommand{\H}{\mrm{H}}
\newcommand{\J}{\mrm{J}}
\newcommand{\K}{\mrm{K}}
\renewcommand{\L}{\mrm{L}}
\newcommand{\Q}{\mrm{Q}}
\newcommand{\R}{\mrm{R}}
\newcommand{\W}{\mrm{W}}
\newcommand{\Z}{\mrm{Z}}
\newcommand{\bbar}{\overline{\mrm{b}}}
\newcommand{\cbar}{\overline{\mrm{c}}}
\newcommand{\dbar}{\overline{\mrm{d}}}
\newcommand{\fbar}{\overline{\mrm{f}}}
\newcommand{\pbar}{\overline{\mrm{p}}}
\newcommand{\qbar}{\overline{\mrm{q}}}
\newcommand{\sbar}{\overline{\mrm{s}}}
\newcommand{\tbar}{\overline{\mrm{t}}}
\newcommand{\ubar}{\overline{\mrm{u}}}
\newcommand{\Fbar}{\overline{\mrm{F}}}
\newcommand{\Qbar}{\overline{\mrm{Q}}}
  
%some useful abbreviations
\newcommand{\Jpsi}{\mrm{J}/\psi}
\newcommand{\ee}{\e^+\e^-}
\newcommand{\ep}{\e\p}
\newcommand{\pp}{\p\p}
\newcommand{\pbarp}{\pbar\p}
\newcommand{\qqbar}{\q\qbar}
\newcommand{\uubar}{\u\ubar}
\newcommand{\ddbar}{\d\dbar}
\newcommand{\ssbar}{\s\sbar}
\newcommand{\gammaZ}{\gamma^* / \Z^0}

%new list environments to replace itemize and enumerate
\newenvironment{Itemize}{\begin{list}{$\bullet$}%
{\setlength{\topsep}{0.2mm}\setlength{\partopsep}{0.2mm}%
\setlength{\itemsep}{0.2mm}\setlength{\parsep}{0.2mm}}}%
{\end{list}}
\newcounter{enumct}
\newenvironment{Enumerate}{\begin{list}{\arabic{enumct}.}%
{\usecounter{enumct}\setlength{\topsep}{0.2mm}%
\setlength{\partopsep}{0.2mm}\setlength{\itemsep}{0.2mm}%
\setlength{\parsep}{0.2mm}}}{\end{list}}
 
%fraction of page devoted to figures
\setcounter{topnumber}{1}
\setcounter{bottomnumber}{1}
\renewcommand{\textfraction}{0.1}
\renewcommand{\topfraction}{0.9}
\renewcommand{\bottomfraction}{0.9}
\renewcommand{\floatpagefraction}{0.8}
 
%one caption and figure, indented but spanning whole page
\newlength{\captivewidth}
\setlength{\captivewidth}{\textwidth}
\addtolength{\captivewidth}{-10mm}
\newcommand{\captive}[1]{\rule{5mm}{0mm}%
\begin{minipage}{\captivewidth}%
\caption[small]{#1}\end{minipage}}
\newcommand{\onefigure}[2]{\begin{figure}[tb]\vspace{#1}
\captive{#2}\end{figure}}

%for indented abstract
\newlength{\abstwidth}
\setlength{\abstwidth}{\textwidth}
\addtolength{\abstwidth}{-1in}

%journals
\def\ap#1#2#3   {{\em Ann. Phys. (NY)} {\bf#1} (#2) #3.}
\def\apj#1#2#3  {{\em Astrophys. J.} {\bf#1} (#2) #3.}
\def\apjl#1#2#3 {{\em Astrophys. J. Lett.} {\bf#1} (#2) #3.}
\def\app#1#2#3  {{\em Acta. Phys. Pol.} {\bf#1} (#2) #3.}
\def\ar#1#2#3   {{\em Ann. Rev. Nucl. Part. Sci.} {\bf#1} (#2) #3.}
\def\cpc#1#2#3  {{\em Computer Phys. Comm.} {\bf#1} (#2) #3.}
\def\err#1#2#3  {{\it Erratum} {\bf#1} (#2) #3.}
\def\ib#1#2#3   {{\it ibid.} {\bf#1} (#2) #3.}
\def\jmp#1#2#3  {{\em J. Math. Phys.} {\bf#1} (#2) #3.}
\def\ijmp#1#2#3 {{\em Int. J. Mod. Phys.} {\bf#1} (#2) #3.}
\def\jetp#1#2#3 {{\em JETP Lett.} {\bf#1} (#2) #3.}
\def\jpg#1#2#3  {{\em J. Phys. G.} {\bf#1} (#2) #3.}
\def\mpl#1#2#3  {{\em Mod. Phys. Lett.} {\bf#1} (#2) #3.}
\def\nat#1#2#3  {{\em Nature (London)} {\bf#1} (#2) #3.}
\def\nc#1#2#3   {{\em Nuovo Cim.} {\bf#1} (#2) #3.}
\def\nim#1#2#3  {{\em Nucl. Instr. Meth.} {\bf#1} (#2) #3.}
\def\np#1#2#3   {{\em Nucl. Phys.} {\bf#1} (#2) #3.}
\def\pcps#1#2#3 {{\em Proc. Cam. Phil. Soc.} {\bf#1} (#2) #3.}
\def\pl#1#2#3   {{\em Phys. Lett.} {\bf#1} (#2) #3.}
\def\prep#1#2#3 {{\em Phys. Rep.} {\bf#1} (#2) #3.}
\def\prev#1#2#3 {{\em Phys. Rev.} {\bf#1} (#2) #3.}
\def\prl#1#2#3  {{\em Phys. Rev. Lett.} {\bf#1} (#2) #3.}
\def\prs#1#2#3  {{\em Proc. Roy. Soc.} {\bf#1} (#2) #3.}
\def\ptp#1#2#3  {{\em Prog. Th. Phys.} {\bf#1} (#2) #3.}
\def\ps#1#2#3   {{\em Physica Scripta} {\bf#1} (#2) #3.}
\def\rmp#1#2#3  {{\em Rev. Mod. Phys.} {\bf#1} (#2) #3.}
\def\rpp#1#2#3  {{\em Rep. Prog. Phys.} {\bf#1} (#2) #3.}
\def\sjnp#1#2#3 {{\em Sov. J. Nucl. Phys.} {\bf#1} (#2) #3.}
\def\spj#1#2#3  {{\em Sov. Phys. JEPT} {\bf#1} (#2) #3.}
\def\spu#1#2#3  {{\em Sov. Phys.-Usp.} {\bf#1} (#2) #3.}
\def\zp#1#2#3   {{\em Zeit. Phys.} {\bf#1} (#2) #3.}
 
\begin{document}
\begin{thebibliography}{99}

\bibitem{main}
G.A.~Schuler and T.~Sj\"ostrand, CERN--TH/95--62 and 
LU TP 95--6, to appear in {\em Zeit. Phys.} {\bf C}.

\bibitem{Gerhard}
G.A.~Schuler, Regensburg preprint TPR--95--15.

\bibitem{gap}
G.A.~Schuler and T.~Sj\"ostrand, \pl{B300}{1993}{169},
~\np{B407}{1993}{539}

\bibitem{Paris}
G.A.~Schuler and T.~Sj\"ostrand, in {\em Two-Photon Physics
from DA$\Phi$NE to LEP200 and Beyond}, World Scientific,
Singapore, 1994, Eds. F.~Kapusta and J.~Parisi, p. 163.

\bibitem{DG}
M.~Drees and K.~Grassie,  \zp{C28}{1985}{451}

\bibitem{LAC} 
H.~Abramowicz, K.~Charchula and A.~Levy, \pl{B269}{1991}{458}.

\bibitem{GRV}
M.~Gl\"{u}ck, E.~Reya and A.~Vogt, \prev{D46}{1992}{1973}

\bibitem{GS}
L.E.~Gordon and J.K.~Storrow, \zp{C56}{1992}{307}

\bibitem{AFG}
P.~Aurenche, M.~Fontannaz and J.-Ph.~Guillet, 
\zp{C64}{1994}{621}

\bibitem{WHIT}
K.~Hagiwara, M.~Tanaka, I.~Watanabe and T.~Izubuchi, 
\prev{D51}{1995}{3197}

\bibitem{FKP}
J.H.~Field, F.~Kapusta and L.~Poggioli, \pl{B181}{1986}{362},
~\zp{C36}{1987}{121}

\end{thebibliography}

 
%set sloppy attitude to line breaks
\sloppy

\pagestyle{empty}

\begin{flushright}
LU TP 95--21 \\
hep-ph/9509264 \\
August 1995
\end{flushright}
 
\vspace{\fill}
 
\begin{center}
{\LARGE\bf Low- and High-Mass Components of}\\[2mm]
{\LARGE\bf the Photon Distribution Functions%
$^{\mbox{\normalsize *}}$}\\[10mm]
{\Large Torbj\"orn Sj\"ostrand} \\[2mm]
Department of Theoretical Physics, University of Lund, \\[1mm]
S\"olvegatan 14A, S-223 62 Lund, Sweden \\[2mm]
and \\[2mm]
{\Large Gerhard A. Schuler} \\[2mm]
Theory Division, CERN \\[1mm]
CH-1211 Geneva 23, Switzerland
\end{center}
 
\vspace{\fill}
 
\begin{center}
{\bf Abstract}\\[2ex]
\vspace{-0.5\baselineskip}
\noindent
\begin{minipage}{\abstwidth}
Four new parton distributions of the
photon are presented, with a description of theory choices,
properties and applications.
\end{minipage}
\end{center}
 
\vspace{\fill}

\noindent
\rule[2mm]{50mm}{0.3mm}\\
$^{\mbox{\normalsize *}}$To appear in the Proceedings of the 
International Europhysics Conference on High Energy Physics, 
Brussels, Belgium, July 27 -- August 2, 1995.\\ 

\clearpage
\pagestyle{plain}
\setcounter{page}{1} 

\section{Introduction}

This is a brief presentation of ref.~1. Related references may be 
found on the non-perturbative constraints on the photon structure 
function\cite{Gerhard}, on photoproduction\cite{gap} and
on $\gamma\gamma$ physics\cite{Paris}. 

There is already available many parton distribution function sets
of the photon\cite{DG,LAC,GRV,GS,AFG,WHIT,FKP}, so why produce more
ones? A few reasons:
\begin{Itemize}
\item The choice of theoretical ansatz is ambiguous, so there is
more room to play than e.g. for the proton.
\item The data are very incomplete and uncertain, so there is a need
for contrasting parametrizations.
\item Compared with previous studies, we put more emphasis on the 
subdivision of the distributions into several components of different
physical nature. This is required for a detailed modelling of
$\gamma\p$ and $\gamma\gamma$ hadronic final states, and for a
sophisticated eikonalization approach to the relation between jet 
and total cross sections. 
\item Our ansatz allows a closed-form extension to the parton 
distributions of a (moderately) virtual photon.
\end{Itemize}  

\section{Physics Assumptions and Fits}

Photons obey a set of inhomogeneous evolution equations, where the 
inhomogeneous term is induced by $\gamma \to \q\qbar$ branchings.
The solution can be written as the sum of two terms,
\begin{equation}
f_a^{\gamma}(x,Q^2) = f_a^{\gamma,\mrm{NP}}(x,Q^2;Q_0^2) 
+ f_a^{\gamma,\mrm{PT}}(x,Q^2;Q_0^2) ~,
\end{equation}
where the former term is a solution to the homogeneous evolution
with a (non-perturbative) input at $Q=Q_0$ and the latter is a
solution to the full inhomogeneous equation with boundary condition
$f_a^{\gamma,\mrm{PT}}(x,Q_0^2;Q_0^2) \equiv 0$. One possible 
physics interpretation is to let $f_a^{\gamma,\mrm{NP}}$ correspond
to $\gamma \leftrightarrow V$ fluctuations, where 
$V = \rho^0, \omega, \phi, \ldots$ is a set of vector mesons,
and let $f_a^{\gamma,\mrm{PT}}$ correspond to perturbative (``anomalous'')
$\gamma \leftrightarrow \q\qbar$ fluctuations. The discrete spectum 
of vector mesons can be combined with the continuous (in virtuality 
$k^2$) spectrum of $\q\qbar$ fluctuations, to give
\begin{displaymath}
f_a^{\gamma}(x,Q^2) =
\sum_V \frac{4\pi\alphaem}{f_V^2} f_a^{\gamma,V}(x,Q^2) \hspace{30mm}
\end{displaymath}
\vspace{-3mm}
\begin{equation}
\mbox{} + \frac{\alphaem}{2\pi} \, \sum_{\q} 2 e_{\q}^2 \, 
\int_{Q_0^2}^{Q^2} \frac{{\d} k^2}{k^2} \, 
f_a^{\gamma,\q\qbar}(x,Q^2;k^2) ~,
\end{equation}
where each component $f^{\gamma,V}$ and $f^{\gamma,\q\qbar}$ obeys a
unit momentum sum rule.

Beyond this fairly general ansatz, a number of choices has to be 
made, as described in the following.

What is $Q_0$? A low scale, $Q_0 \approx 0.6$~GeV, is favoured if
the $V$ states above are to be associated with the lowest-lying
resonances only. Then one expects $Q_0 \sim m_{\rho}/2$--$m_{\rho}$.
Furthermore, with this $Q_0$ one obtains a reasonable description of 
the total $\gamma\p$ cross
section, and continuity e.g. in the primordial $k_{\perp}$ spectrum.
Against this choice speaks worries that perturbation theory may not
be valid at such low $Q$, or at least that higher-twist terms appear
in addition to the standard ones. Alternatively one could 
therefore pick a larger value, $Q_0 \approx 2$~GeV, where these
worries are absent. One then needs to include also higher resonances
in the vector-meson sector, which adds some arbitrariness, and one 
can no longer compare with low-$Q$ data. We have chosen to prepare 
sets for both these (extreme) alternatives.

How handle the direct contribution? Unlike the p, the $\gamma$ has a 
direct component where the photon acts as an unresolved probe.
In the definition of $F_2^{\gamma}$ this adds a component 
$C^{\gamma}$, symbolically
\begin{equation}
F_2^{\gamma}(x,Q^2) = \sum_{\q} e_{\q}^2 \left[ f_{\q}^{\gamma} + 
f_{\qbar}^{\gamma} \right] \otimes C_{\q} + 
f_{\g}^{\gamma} \otimes C_{\g} + C^{\gamma} ~.
\end{equation}
Since $C^{\gamma} \equiv 0$ in leading order, and since we stay with 
leading-order fits, it is permissible to neglect this complication.
Numerically, however, it makes a non-negligible difference. We 
therefore make two kinds of fits, one DIS type with $C^{\gamma} = 0$
and one $\overline{\mrm{MS}}$ type including the universal part
of $C^{\gamma}$\cite{AFG}.

How deal with heavy flavours, i.e. mainly charm? When jet production
is studied for real incoming photons, the standard evolution approach 
is reasonable, but with a lower cut-off $Q_0 \approx m_{\c}$ for
$\gamma \to \c\cbar$. Moving to deep inelastic scattering, 
$\e\gamma \to \e X$, there is an extra kinematical constraint:
$W^2 = Q^2 (1-x)/x > 4 m_{\c}^2$. It is here better to use the
``Bethe-Heitler'' cross section for $\gamma^* \gamma^* \to \c\cbar$.
Therefore two kinds of output is provided. The parton distributions 
are calculated as the sum of a vector-meson part and an anomalous part
including all five flavours, while $F_2^{\gamma}$ is calculated
separately from the sum of the same vector-meson part, an anomalous
part and possibly a $C^{\gamma}$ part now only covering the three 
light flavours, and a Bethe-Heitler part for c and b.

Should $\rho^0$ and $\omega$ be added coherently or incoherently?
In a coherent mixture, $\u\ubar : \d\dbar = 4 : 1$ at $Q_0$, while 
the incoherent mixture gives $1 : 1$. We argue for
coherence at the short distances probed by parton distributions. 
This contrasts with long-distance processes, such as 
$\e\gamma \to \e V$.

What is $\Lambda_{\mrm{QCD}}$? The data are not good enough to allow
a precise determination. Therefore we use a fixed value 
$\Lambda^{(4)} = 200$~MeV, in agreement with conventional 
results for proton distributions.

\begin{figure}[t]
\begin{center}
\mbox{\epsfig{figure=lutp95211.eps,width=150mm}}
\end{center}
\caption{Subdivision of the full $F_2^{\gamma}$ parametrization by
component, compared with data at $Q^2 \approx 5.2$~GeV$^2$.
The total $F_2^{\gamma}$ (except for an overall factor 
$\alpha_{\mbox{\tiny em}}$) is shown by the full curve. 
The lowest dashed curve gives the VMD contribution, and the next 
lowest the sum of VMD and anomalous ones. The third dashed curve,
which coincides with the full curve for the DIS fits, gives
the sum of VMD, anomalous and Bethe--Heitler terms. For the 
$\overline{\mrm{MS}}$ fits, the full curve additionally contains the
contribution of the $C^{\gamma}$ term. Note that this last term is 
negative at large $x$.}
\end{figure}

In total, four distributions are presented\cite{main}, based on fits 
to available data:
\begin{Itemize}
\item SaS 1D, with $Q_0 = 0.6$~GeV and in the DIS scheme.
\item SaS 1M, with $Q_0 = 0.6$~GeV and in the $\overline{\mrm{MS}}$
scheme.
\item SaS 2D, with $Q_0 = 2$~GeV and in the DIS scheme.
\item SaS 2M, with $Q_0 = 2$~GeV and in the $\overline{\mrm{MS}}$ 
scheme.
\end{Itemize}
Fig.~1 compares these distributions at one $Q^2$ value. 

The code for the parametrizations is freely available, 
e.g. on the web:\\
{\tt http://thep.lu.se/tf2/staff/torbjorn/lsasgam}

The parametrizations above can be used also for photons with
a virtuality $P^2 \neq 0$. This is done by introducing dipole factors 
$m_V^4/(m_V^2 + P^2)^2$ for the VMD components, by changing the lower
cut-off of the anomalous integral from $Q_0^2$ to $\max(Q_0^2, P^2)$,
and by using the proper off-shell expressions for Bethe-Heitler 
cross sections etc.  

\section{Summary}

Above are presented four new sets of parton distributions and
structure functions of the photon\cite{main}. Owing to the subdivision 
of the distributions into different physics components, it is now
possible to construct detailed models of $\gamma\p$ and 
$\gamma\gamma$ events. These models are presented 
elsewhere\cite{gap,Paris}, with further work in progress.



\end{document}


